

Received November 11, 2020, accepted December 6, 2020, date of publication December 11, 2020, date of current version December 29, 2020.

Digital Object Identifier 10.1109/ACCESS.2020.3044040

A Survey of Technologies for Unmanned Merchant Ships

JIA WANG^{1,2}, YANG XIAO¹⁰³, (Fellow, IEEE), TIESHAN LI^{101,4}, (Senior Member, IEEE), AND C. L. PHILIP CHEN^{101,5}, (Fellow, IEEE)

¹Navigation College, Dalian Maritime University, Dalian 116026, China

²Experimental Instrument Center, Dalian Polytechnic University, Dalian 116034, China

³Department of Computer Science, The University of Alabama, Tuscaloosa, AL 35487-0290, USA

⁴School of Automation Engineering, University of Electronic Science and Technology of China, Chengdu 610054, China

⁵School of Computer Science and Engineering, South China University of Technology, Guangzhou 510006, China

Corresponding author: Yang Xiao (yangxiao@ieee.org)

This work was supported in part by the National Natural Science Foundation of China under Grant 51939001 and Grant 61976033, and in part by the Science and Technology Innovation Funds of Dalian under Grant 2018J11CY022.

ABSTRACT Unmanned merchant ships are ships that carry goods and engage in commercial activities without manual operations on board. In contrast to traditional merchant ships, unmanned merchant ships have great advantages in economy, society, and safety. However, we observe that commercial applications of unmanned ships are still at an exploratory stage. We believe that unmanned merchant ships will be widely adopted and popular in the near future. There is a need to have a good survey of current technologies, foundation, and obstacles of implementation of future unmanned merchant ships. Such a survey is beneficial to ship builders, researchers, owners, and students. Therefore, in this paper, we present a comprehensive survey of ship structure and technology, navigation, automation, algorithms, communication, Internet of Things (IOT), etc. We introduce traditional ships and analyze the role of crew and navigation operations on board. We summarize classifications, benefits, and core technologies of IOT for unmanned merchant ships. Moreover, we introduce intelligent awareness, data fusion, applications, and E-navigation for unmanned merchant ships. We also present several future research directions at the end.

INDEX TERMS Merchant ships, unmanned ships, Internet of Things (IoT), algorithms, route planning, collision avoidance, sensors, radio frequency identification (RFID), Zigbee, wireless sensor network (WSN), very high frequency (VHF), E-navigation, intelligent awareness, data fusion, communication, Ship Industry 4.0 (SI4), smart ocean, AI, big data.

I. INTRODUCTION

Merchant shipping is a general term for ships engaged in none-fishing commercial activities [1]. With 90 percent of the world trade carried by merchant ships at sea, merchant shipping is one of the most important modes of transportation [2]. A merchant ship is a watercraft that transports cargo or carries passengers for hire and is different from entertainment ships and navy ships in its use and purpose [3]. Navigation of a traditional merchant ship mainly depends on crew operation so that navigation safety is easily affected by human factors such as fatigue, attention, and emotion of the crew. The number of accidents caused by human error is in the range

The associate editor coordinating the review of this manuscript and approving it for publication was Yuan Tian¹⁰.

of 70-95% [4]. Therefore, ship owners have a strong desire for such ship types which are intelligent, green, safe, and efficient.

Unmanned ships are ships that operate on water surface without a crew [5]. We can define an unmanned merchant ship as a merchant ship without a crew on board, which use advanced methods/algorithms, intelligent awareness, data fusion, communications, Internet of Things (IOT), control technologies, and other technologies to implement automatic operation, navigation, and berthing. The ship can be intelligently operated based on navigation technologies, computer technologies, automatic control technologies during navigating, managing, maintaining, transporting cargo so that the merchant ship has more advantages in terms of safety, society, and economy.

Unmanned merchant ships have or will have great advantages in efficiency, cost effectiveness, environmental friendliness, work safety, and family friendliness [6]. We list four major advantages of unmanned merchant ships as follows. First, merchant unmanned ships are independent of crew and are operated mainly by expert decision-making systems, artificial intelligence systems, and remote control systems in shore-based control centers to reduce the influence of human factors on safety of ship navigation. According to International Union of Marine Insurance (IUMI) [7] and the authors in [4], much of this safety improvement has to do with improvements in automation systems supporting human monitoring and decision-making. Second, unmanned merchant ships can save energy and improve economic benefits. Onboard labor costs accounts for 31-36% in the total ship operation cost [6]. The tests in [6], [8] show that slow sailing of ships can save fuel and reduce pollution emissions. Furthermore, because there are no people on unmanned ships, human cost can be reduced since people will have to do the maintenance, supervision, contingency handling, etc. On the premise of ensuring timely arrival, unmanned ships can sail slowly as much as possible to save fuel. Third, compared with traditional merchant ships, unmanned merchant ships do not need to consider all crew related facilities such as lifesaving, fire-fighting, pollution prevention and life, saving weight, space, and labor costs, and can carry more cargo [9]. Fourth, for unmanned merchant ships, pirates cannot threaten hostages so that this makes unmanned merchant ships relatively safer.

Unmanned merchant ships must be smart and autonomous to implement the goal of replacing crew handing. Several key technologies for smart ships include information sensing, communication and navigation, energy efficiency control, route planning, status monitoring and fault diagnosis, distress warning and rescue, and autonomous navigation [10]. The report [11] shows three vital elements that will make autonomous shipping a reality: sensor fusion, control algorithms, and communications. Therefore, we can see that Internet of Things (IoT), algorithms, and communication are among major factors of unmanned merchant ships with the reasons as follows. Firstly, IoT extends the Internet as the core and foundation to the information exchange and communication between any things to implement the functions of intelligent identification, location, tracking, supervision, etc [12]-[14]. The major technologies of IoT include Radio frequency identification (RFID), Zigbee, Wireless Sensor Network (WSN) [15], etc. Secondly, there are many algorithms which can find the optimal routes and avoid collisions for unmanned ships [16]–[19]. Thirdly, communications of unmanned ships include Controller Area Network(CAN) bus, WSN, Zigbee, 3G/4G/5G, satellite communication, Very high frequency (VHF), etc. These communications will connect devices of unmanned ships together.

To implement commercialized unmanned merchant ships and promote research in all aspects, there is a need to have a very systematic survey about the state of art of development and core technologies of unmanned merchant ships. Such a review will be very useful for ship builders, ship owners, researchers, and students. In this paper, we will provide such a comprehensive review of the state of art of core technologies in the field of unmanned merchant ships. The contributions of this paper are listed as follows:

- To the best of our knowledge, this paper is the first paper in the literature to present a comprehensive survey and tutorial on major factors of unmanned merchant ships.
- We survey basic structure of ships, principle of ship handing, navigation instruments, and responsibility of crew of traditional ships. This tutorial provides details of shipping technologies to the readers for their future research in these fields.
- We survey existing papers of unmanned ships combining knowledge in the field of navigation, computer science, communication, control, and automation. We classify unmanned ships and list the benefits. Furthermore, We summarize the existing core technologies for unmanned ships and analyze the advantages and disadvantages.
- We summarize intelligent awareness, data fusion, communication, IOT applications, and E-navigation for unmanned merchant ships. Although there are not many research work available in the literature about IOT for shipping, the paper shows the great and promising area of IOT for future shipping industries. Our paper is the first paper to try to survey this important aspect of IOT on shipping so that we believe that our paper will inspire many researches to work along this line.
- We discuss the challenges and further research for unmanned merchant ships from the aspects of law, communication, cyber security, IOT application onboard, intelligent auxiliary equipment, automatic control, artificial intelligence and big data, ship industry 4.0, and smart ocean.

This paper is organized as follows. Section II introduces basic information of a ship and duties of personnel on board. Section III provides an in-depth review of current research of unmanned ships. Section IV presents application of IoT and other technologies in unmanned merchant ships. Section V discusses challenges and further research of IoT for unmanned merchant ships. Finally we conclude this paper in Section VI.

II. MECHANISMS OF A MERCHANT SHIP

In order to better understand unmanned merchant ships, we first introduce general merchant ships from structure, usage, handling principles, navigation instruments, and the duties of onboard staff [20], [21]. The photos in this section were taken by us for a ship called "Yukun" of Dalian Maritime University. We use the ship as an example to introduce shipping functionalities with the parameters shown in Table. 1 and explained as follows:

Length	116.00 m
Breadth	18.00 m
Depth	11.10 m
Designed draft	5.40 m
Main engine	4440kW * 173rpm
Service speed	16.7 knots
Endurance	10000 n.miles
Complement	236 р
Classification	CSA Training Ship, Ice Class B, CSM AUT-0 SCM

TABLE 1. Parameters of the ship "Yukun".

- CSA It means that the ship's structure and equipment are in full compliance with CCS (China Classification Society) specifications and are suitable for navigation in an infinite navigation area.
- CSM It means that the manufacturing and installation of the ship's propulsion machinery and auxiliary machinery for important purposes conform to the provisions of CCS specifications and are suitable for navigation in the infinite navigation area.
- AUT-0 The degree of automation of the engine room can be unattended during navigation.
- SCM Propeller condition monitoring.



FIGURE 1. A model of navigation ship, called Yukun, Dalian Maritime University; here we show a model instead of the real ship.

A. BASIC STRUCTURE OF A SHIP

A merchant ship consists of a main hull, a superstructure, and various equipment, shown in Fig. 1. The main hull is a watertight hollow structure composed of upper deck, bottom, broadside, deck, fore and aft, bulkhead, etc. The superstructure includes forecastle, navigating bridge, poop and deckhouse. Apart from cabins with various functions in the superstructure, the main hull is separated into several cabins by each deck and bulkhead, including an engine room, a cargo room, a ballast tank, a deep tank, a fuel oil tank, a lubricating oil tank, a fresh water tank, a slop tank, and a caisson in this example. The basic structure of a traditional ship is shown in Fig. 2, in which there are various



FIGURE 2. Main structure of a traditional ship.

equipments on a traditional ship, including an echo sounder, a DGPS, a Radar, a speed log, an Electronic Chart Display and Information System (ECDIS), an Automatic Identification Systems (AIS), an autopilot, a compass, a satellite radio, etc. The shipboard equipment and navigation instruments are introduced in subsection II-C. Navigation of a merchant ship relies on supporting equipment, including a main machinery, an auxiliary machinery, auxiliary equipment, electricity, various pipes, deck equipment, safety equipment, communication equipment, navigation equipment, and living supporting equipment.

The engine room refers to the machinery spaces of a ship, where a main engine, an auxiliary engine, and other auxiliary equipments are located [22], shown in Fig. 2. According to the nature of fuel, the burning place, and the working mode, the main machinery of the ship can be classified to a steam engine, an internal combustion engine, a nuclear power engine, and motors. In the 20th century, most of ships sail on water by propelling devices of mechanical propulsion [20]. With the development of ship technologies, there are many new technologies for ship propulsion, such as water jet propulsion, azimuth propulsion, pump jet propulsion, and air cushion propulsion.

A steering gear is a main equipment to control a ship, and its function is to maintain a required course, changing original course, and carry out cyclic motion. The steering gear is usually positioned at the stern to produce a large turning moment to turn the ship, shown in Fig. 3. The steering gear are housed in the steering gear cabin on the stern peak deck platform. The control system of the steering gear is installed in the driving room, and the steering order is transmitted to the steering gear through the electric or hydraulic control system from the cab to control its action.

According to usage, piping system can be classified to bilge system, ballast pipe, ventilating system, fire extinguishing system, domestic water supply system, deck scupper system, and sanitary water system, etc.



FIGURE 3. Basic structure of a ship.

An anchor equipment is an important equipment in deck equipment, which is used for ship anchoring, auxiliary control when the ship leans away from the wharf, deceleration of ship speed when sailing in narrow channel, and so on, shown in Fig. 3. An windlass is a mechanical device for dropping and heaving up an anchor, mainly an electric anchor and a hydraulic anchor, located at the bow, which can be remotely operated in the cab, shown in Fig. 3.

B. BASIC PRINCIPLES OF SHIP HANDLING

Ship handling refers to the relative motion of the hull, propeller, and rudder in water to produce hydrodynamic force to keep or change the horizontal motion of the ship. According to the helmsman's or conning officer's intention, the ship can maintain or change its speed, course, and position.

In order to reduce the work-related stress of the helmsman, it can correct the yaw in time and keep the ship on the course for a long time. Generally, a big ship at sea is installed with an automatic course-keeping control system, called autopilot. An autopilot is basically composed of main mechanisms such as automatic detection of course deviation, signal comparison, signal amplification, actuator, and feedback. When a ship is in the autopilot state, a gyrocompass can detect whether the ship is off-course.

A compass is widely used in all kinds of ships for navigation, shown in Fig. 4(a). A gyrocompass will detect the yaw angle ϕ when the ship off-course. An automatic steering transmitter outputs AC voltage proportional to a yaw angle, through phase-sensitive rectifying into different polarity DC voltage U_{ϕ} . The voltage signal is transmitted to a comparison circuit, which starts motors to turn the rudder left or right through an amplifier and a switch circuit. Many ships adopt adaptive autopilot (autopilot for short) to reduce the turning times and resistance of rudder, to keep the ship in the original course, and to save fuel consumption.

An automatic navigation (or navpilot) is a computer-based automatic navigation control system. It is based on autopilot and connected with the integrated navigator or ship position receiver to provide the ship with reliable and accurate course control. However, manual steering shall be adopted instead in case of severe weather or heavy traffic.

A ship must anchor while waiting for berthing, anchorage operation, and wind shelter. The handing of an anchor is usually in the charge of a chief officer. The chief officer carries out the anchoring order issued by the bridge. Meanwhile, the chief officer must reports to the bridge the status and execution of the anchor chain during anchoring.

C. NAVIGATION INSTRUMENTS

With the development of large-scale ship tonnage, high-speed movement, and intensive navigation, it is difficult to cope with complicated navigation environment with simple outlook and experience. We must use the navigational instruments widely to ensure safety of ship navigation. Several navigation instruments are widely used and described below.

As a navigation pointing instrument, the magnetic needle of a magnetic compass always points to the magnetic North Pole of the earth by the mutual attraction of geomagnetic field. A magnetic compass can indicate the course and the position of a ship, and help a ship to implement positioning and navigation functions. A magnetic compass and a gyrocompass complement each other to jointly guarantee the safe navigation of the ship. In recent years, A kind of solidstate electronic magnetic compass has been developed and a magnetic sensor is core. It uses a magnetic sensor to sense the external magnetic field and convert the induction of external magnetic field into an electric signal. At present, there are Hall compasses, flux gate compasses, magnetoresistance compasses, Giant magnetoresistive compasses, etc.

A marine echo sounder, which is a kind of acoustic navigation instrument, utilizes the physical characteristics of ultrasonic wave propagation in water for measuring water depth, shown in Fig. 4 (b). The principle of the marine echo sounder is to determine the depth of water by measuring the time interval between the ultrasonic wave emitted and the wave received after reflection from the bottom.

A speed log is a kind of navigation instrument to measure ship speed and accumulated voyage, shown in Fig. 4 (f). An electromagnetic log used the electromagnetic induction principle to measure the speed and voyage of the ship relative to the water. It is composed of an electromagnetic sensor, an amplifier, and an indicator. It has the advantages of good speed detection linearity, large range, high precision, low cost, and easy to use.

A Global Positioning System (GPS) satellite navigator is a GPS receiver specially used for positioning and navigation. Difference Global Positioning System (DGPS) is to add correction signal on the basis of GPS to improve the accuracy of GPS, shown in Fig. 4 (c). It can be used for route point



(i) An integrated bridge system

FIGURE 4. Navigation instruments in bridge of a ship.

navigation, route navigation, track line plotting, fixed-point navigation, alarm, etc. The Coarse/Acquisition (CA) code is more commonly used for single frequency GPS satellite navigator. It is a kind of pseudo-random noise code used to broadcast rough ranging and fast acquisition of the precise code via GPS satellite.

Radar is short for Radio detection and ranging. A marine radar is X band or S band radar on ships that detects targets by emitting electromagnetic waves and receiving the targets' reflected echo. It usually used to detect surface ships, obstacles, lands, maritime distress, and sea clutter for collision avoidance, location, and navigation [23], shown in Fig. 4(d). A radar consists of timer, transmitter, transceiver switch, antenna, receiver, display, and radar power supply. Radar/APRA (Automatic Radar Plotting Aid), Electronic Chart Display and Information System (ECDIS), GPS/DGPS, and autopilot constitute an automatic bridge system, which is currently important navigation systems on ships.

An ECDIS is an advanced nautical navigation information system used for ships [24], shown in Fig. 4 (e). It displays the position of the ship and its intended movement in relation to the navigational characteristics on the chart [5]. An ECDIS combines satellite position, GPS, sensors, radar, Navtex, Automatic Identification Systems (AIS), depth sounders, and other data with a sophisticated electronic database containing chart information for decision aid [25], where Navtex is an automated service for navigational and meteorological warnings and forecasts.

An AIS is a navigation auxiliary system which can automatically broadcast and receive the static information, dynamic information, voyage information, and safety information of the ship and surrounding ships through Very high frequency (VHF) band and Time-division multiple access (TDMA) technology. An AIS implements ship identification, monitoring, and communication, shown in Fig. 4(g). Its cores include satellite positioning technologies, digital communication technologies, information processing technologies, and computer network technologies. An AIS receives the position information from a GPS/DGPS/GNSS(Global Navigation Satellite System) receiver, the course information of a gyrocompass, and the speed information of the log through the interface circuit. The information processor processes and sends the ship's static

TABLE 2. The responsibilities of the crew on board [26].

Post	Responsibility
captain	The captain is the leader of the ship and responsible to both the ship owner (ship company) and those functions such as the safety production, navigation command, administrative management, technical business, and foreign-related work of the ship.
chief officer	The chief officer is the captain's chief assistant and the head of the deck department, in charge of the daily work of the deck department. In addition to navigation watch, he/she is in charge of cargo stowage, loading and unloading, handover and transportation management, as well as the maintenance of the deck department.
second officer	The second officer, under the leadership of the captain and the chief officer, performs the duty of navigation and berthing, and is in charge of the bridge equipment, including all kinds of radio navigational instruments, meteorological instruments, steering instruments, chronometers and ship clocks, compass, national flag, signal flag, signal lamp, model, charts, and other navigational books and materials.
third officer	The third officer shall, under the leadership of the captain and the first officer, performs the duty of navigation and berthing, and is in charge of life-saving and fire-fighting equipment.
radio operator	Under the leadership of the captain, the radio operator is on duty on time and operates all kinds of radio communication equipment correctly. The radio operator must complete the task of maintenance and communication assigned by the captain, and make records.
boatswain	The boatswain, under the chief officer, organizes and leads carpenter and crew in their work.
carpenter	The carpenter is responsible for carpentry and related work under the leadership of the chief officer and the boatswain.
cook	The cook is responsible for cooking range, warehouse, and catering.
chief engineer officer	The chief engineer, under the leadership of the captain, is the chief technical officer in charge of the mechanical, power and electrical (except radio instruments) equipment of the ship.
first engineer officer	The first engineer is the chief assistant of the chief engineer. Under the leadership of the chief engineer, the first engineer is responsible for leading the staff of the engine department to manage, operate, and overhaul electromechanical equipment to ensure the correct implementation of various rules and regulations of the engine department.
second engineer officer	The second engineer, under the leadership of the chief engineer and the first engineer, is responsible for the management of the generator and the mechanical equipment for its service, some auxiliary engines in the engine room, and other equipment designated by the chief engineer. He / She is responsible for loading fuel oil (refueling oil), measuring and recording fuel oil.
third engineer officer	Under the leadership of the chief engineer and the first engineer, the third engineer is responsible for the management of deck machinery, pump rooms, lifeboats, emergency fire pumps, air conditioners, auxiliary boilers and their auxiliary equipment, and some auxiliary machines in the engine room, as well as other auxiliary machines and equipment designated by the chief engineer. For ships without electrical engineers, the third engineer is responsible for managing the electrical equipment.
master mechanic officer	The master mechanic shall have the ability to manage, operate, and repair. Under the leadership of the first engineer, the master mechanic is responsible for organizing and arranging the shift of the machinists and the cleaning and daily maintenance of the machinery, furnace and pump compartment.
mechanic	Under the leadership of the chief engineer and the first engineer, and under the direct leadership and arrangement of the master mechanic, the mechanic shall independently conduct the overhaul, maintenance and cleaning of machinery, electricity, boiler equipment, pipe systems and valves, and is on duty within the prescribed time.

information and dynamic navigation information through a VHF transceiver. At the same time, it also receives the navigation data of surrounding ships, and displays it on the information monitor.

A ship voyage data recorder, which is an equipment in a safe and recoverable way in real time, records the information about the position, dynamics, physical condition, handing and navigation data of a ship in the period before and after the accident, shown in Fig. 4h. It includes data processor, microphone set, sensor interface and signal processing circuit, data protection module, alarm indicator, power supply, and data playback equipment. Integrated Bridge System (IBS) bases on Integrated Navigation System (INS). IBS integrates the navigation instrument above and combines radar, electronic chart, AIS, autopilot, and various types of navigation and ship handing device, shown in Fig. 4(i). It has integrated navigation, ship handing, automatic collision avoidance, integrated information display, communication, and navigation management control functions.

D. THE RESPONSIBILITIES OF THE CREW ON BOARD

We review crew and their job responsibilities in this subsection since these are useful for unmanned merchant ships.



FIGURE 5. General unmanned ship architecture [31].

The responsibilities of these men on unmanned merchant ships are mainly replaced by mechanical, electronic, and computer technology. Besides the captain, there are also deck department and engine department personnel on board, as shown in Table 2 [26]. The deck department is mainly responsible for the navigation, hull maintenance, cargo stowage, loading and unloading equipment, and the care of cargo during the voyage. The deck department includes a chief officer, a second officer, a third officer, boatswains, carpenters, crew, helmsman, and cooks. The engine department is mainly responsible for the management, use, and maintenance of the main engine, boiler, auxiliary machinery, and various electromechanical equipment, as well as the management and maintenance of the whole ship's power system. The engine department mainly includes a chief engineer, a first engineer, a second engineer, a third engineer, a master mechanic, coppersmith, and mechanics.

In future unmanned merchant ships, most of these job responsibilities will be replaced by machines and some of them will be no loner needed, particularly for those serving crew such as cooks.

III. CURRENT STATUS OF UNMANNED SHIPS

In this section, we introduce unmanned ships. First, we classify the existing unmanned ships based on automaticity and remote control. We analyze the advantages of unmanned ships from the impacts of unmanned ship development on safety, economy, society, and environment, etc. Second, we research the history and current status of unmanned ships. At last, we summarize the core technologies of unmanned ships.

A. CLASSIFICATION AND BENEFITS OF UNMANNED SHIPS

Unmanned ships are ships that operate on the surface of the water without a crew [5]. Carderock Laboratory uses the following method of grouping Unmanned Surface Vehicles (USVs) by displacement [27]: small(<1t), medium(<100t), large(<1000t), and extra large(>1000t), where displacement refers to the total weight of a ship under

a given condition and usually expressed in tonnage, abbreviated as t. We focus on large and extra large displacement USVs as unmanned ships in this paper. At present, there are many researches on small and medium USVs [28]–[30]. Research on unmanned ships can draw lessons from the existing research results of small and medium USVs.

A system architecture for an unmanned ship includes an instrument layer, a process layer, a integrated ship control layer, and a general ship layer, shown in Fig. 5 [31]. We introduce Fig. 5 in detail as follows. Generally, a gyrocompass provides compass signals for radars and an electronic chart system integrating GPS, AIS, Radar, and other information. According to sensor information, actuators perform corresponding operations on engines. Ship control system integration mainly consists of three parts: energy management, performance monitoring, and integrated ship control database and access. The general ship layer includes two parts. The first part is the network used by crew and passengers for living accommodation. The other part is the administrative network, including safety management, reporting applications, and maintenance. The communication of unmanned ships mainly uses satellite and 3G/4G/5G networks. Unmanned ships are monitored and remotely controlled by Shore Control Centers (SCCs) when necessary [32]. To minimize operators' load, unmanned ships are able to operate autonomously without operators for most of the time [31].

International Maritime Organization (IMO) classifies the automation of autonomous surface ships into several levels, shown in Table 3 [33]:

- Automated manned ships: Functional systems of ships are operated and controlled by the crew on broad. Some operations can be automated.
- Remote-control ships with crew on board: There is a certain number of crew on board. But ship's operations and decisions mainly rely on remote control through communication.
- Remote-control ships without crew on board: Control and operation of the ships take place in another place and there is no crew on board.

level	remote con- trol	crew on board	autonomous
automated manned ships	no	yes	no
remote-control ships with crew on board	yes	yes	no
remote-control ships without crew on board	yes	no	no
completely unmanned automatic ships	yes	no	yes

TABLE 3. IMO classification of the automation level of autonomous surface ships [33].

• Completely unmanned automatic ships: Decision control systems of ships are able to make decisions, to react on automatic pilot, and to act on navigation control system independently.

An unmanned ship may have a certain number of people on board, but it does not need to be performed or supervised by the people. An unmanned ship may have operational and technical problems in crowded waters. In addition, an unmanned ship must comply with the laws of the country where the port is located. Therefore, the unmanned ship should be undertaken with a crew on board when it enters and leaves the port [31]. The crew can board and disembark via shuttle boats or helicopters after the crew has completed the operation within the specified voyage.

A remote-control ship with a certain number of crew on board has many advanced automation systems. It can perform some demanding operations without crew interaction, such as dynamic positioning or automatic berthing. A remote-control ship may need remote control operations in case of accidents or completion of operations. When a manual operation is required, the crew of Shore Control Centers (SCCs) can intervene and remotely control the unmanned ship, and the crew in the bridge can directly control it [34].

A completely unmanned automatic ship can handle all situations when it is sailing and berthing. This implies that there is no SCC or any crew in the bridge at all. However, we believe that a completely unmanned ship is not likely to be implemented in near future.

Next, we try to make a distinction between an autonomous ship and an unmanned ship. They are different in some degree. Autonomous operation and remote control are the important characteristics of unmanned ships. We can define different types of ship autonomy based on operational autonomy levels [32]. An autonomous ship can perform a series of pre-designed operations, whether or not the crew is present [32]. The report [35] shows that future advanced control systems, decision support systems, and wireless communication technologies will enable autonomous ships on SCCs to have the same monitoring and control functions as those on ships.

Many researchers work on navigation control for unmanned ships, as illustrated below. The authors in [36] design a navigation control for an unmanned ship explained as follows. The navigation control is composed of a ship's intelligent system, a Narrow Band Internet of Things (NB-IoT) based station, and a shore-based support center. The authors establish a communication link between the ship and SCCs through NB-IoT and implement a system which enables the unmanned ship of the actual track approximating the expected track. The NB-IoT communication link is supported by the SARA-N2 module with maximum speeds of 227kbps downlink and 21kbps uplink. SARA-N2 is released by u-blox, Switzerland on June 28, 2016 [37]. It is the world's first mobile wireless module following Release 13 issued by the Third GeneraTIon Partnership Project (3GPP). It has the advantages of ultra low power consumption delivering 10+ years battery life, extended temperature range of -40 to $+85^{\circ}$ C, easy migration between u-blox 2G, 3G, and 4G modules [38]. The NB-IoT based station of the telecom operator enables real-time data sharing between the ship and the shore. Users of the unmanned ship [36] can monitor the ship's position and movement in real-time and control the ship when necessary at SCCs.

Next, we explain the benefits of unmanned ships. Much of world trade depends on sea transportation. There is 90% of world trade carried by sea, worth about \$375 billion [39]. Unmanned ships can reduce the operation cost, eliminate the on-board crew cost, reduce the risk associated with human errors, and reduce the threat of crew from pirates [4]. We introduce the benefits of unmanned ships from social, economic, and security aspects [40].

First of all, unmanned ships can reduce human costs. When a ship is sailing at the sea for a long time, the crew may feel lonely in a closed environment and this may reduce the attractiveness of the job. It is hard to recruit sufficient qualified crew-members. The crew may encounter natural influences such as wind, waves, fog, and typhoon, which require realtime operation and responses during the voyage. The change of geographical location causes the seasonal climate change and jet lag of voyage and this will affect the mood and psychology of the crew [41].

Secondly, unmanned ships can reduce human costs. The demand for ocean-going trades is increasing. More and more unmanned ships are needed in the future. With the increase of manpower cost, the role of unmanned ships in reducing costs is becoming more and more obvious. According to statistics, the one-day crew cost of a large container ship accounts for 44% of the total operating cost, about \$3,299 [39]. Furthermore, life and life-saving equipments on the ships are for the service of the crew. Unmanned ships do not need these facilities and this can reduce weight by 5% and save 12% to 15% on fuel [9], [42].

Thirdly, navigation at sea is a high-risk industry. Maritime safety is the most important thing in ocean transportation. The main causes of marine accidents are closely related to people's sense of responsibility, skills, experience, physical and mental conditions, and strain capacity. Professionals agree that "human error" is the cause of between 70% and 95%

of shipping accidents [4]. Unmanned ships can reduce risk associated with human errors. Unmanned ships integrate the data and information of radars, High-Definition (HD) infrared cameras, and a variety of sensors, relying on artificial intelligence, remote control, and data fusion to ensure the security and accuracy of the ship operation. Furthermore, unmanned ships can effectively defend against pirates' attacks. If unmanned ships don't have easy access, pirates will have a hard time entering inside of the ships [43]. There is no hostage risk with an unmanned ship because there is no crew on board. Even boarding the ship, access to the control system is difficult. Remote control centers can allow unmanned ships to anchor or move at low speeds, making the ships easy for rescue ships to come in to help [44].

B. THE STATE OF ART OF UNMANNED SHIPS

While unmanned ships date back at least to World War II, a lot of unmanned ship projects appears in the 1990s [45]. United States Navy developed unmanned ships and they were mainly used for mine sweeping, intelligence gathering, and other dangerous tasks around the 1990s [45]. In recent years, USVs are widely used in military, environmental, and robotic research applications [29]. Next, we illustrate some practical applications of unmanned ships.

United States Navy focuses on the development of unmanned ships. In June 2003 [45], the Office of Naval Research (ONR) provided funding to the US Naval Facilities Engineering Support Center (NFESC) for developing a small USV called SeaFox. It is a semi-autonomous wireguided mine disposal vehicle. Shore-based operators rely on video images transmitted over fiber-optic cables to remotely control SeaFox and destroy mines with integrated shaped charge [46]. Maritime Applied Physics Corporation (MARP) built a high-speed USV, which is a 10 meter hydrofoil with a top speed of more than 40 knots in a rough sea [47]. The ONR tested this high-speed USV in 2005 [47]. In May 2016 [48], United States began testing "Sea Hunter" which is the largest unmanned warship of the world at that time. Sea Hunter can travel at up to 27 knots and navigate without human intervention through the most extreme water conditions, including Sea State 5 [49].

Besides United States, several other countries also develop USVs. Japanese company Yamaha developed two USVs in 2004 [50]: UMV-H and UMV-O. The model UMV-H is a high-speed craft of 4.44m in length equipped with 90 kw and 40 knots, and the UMV-O is a displacement hull for going deep in the ocean [51], [52]. The Canadian company International Submarine Engineering Ltd (ISE) had worked on USVs for more than 35 years since 1984 [45]. They converted a manned ship into a USV using Tactical Controller (TC) Kit developed by ISE. Israeli company Elbit Systems unveiled Stingray, one USV, at the 2005 IDEF-05 exhibition in Turkey [53]. Silver Marlin developed by Elbit Systems is an autonomous and medium-sized USV featuring autonomous obstacle avoidance sensors and controllers [54]. In addition to the military role of unmanned ships, many attentions focused on industry commercialization now. Data transmission between ships and SCC is one of the core technologies for unmanned ships. In September 2015, Rolls-Royce led a new 6.6 million euro project for autonomous ships [56]. This project was built on existing ship-to-ship and ship-to-shore communication platforms and was the world's first hybrid Ka/L-band mobile satellite system. It was to provide solutions for the next generation of advanced ships and tested in 2017 [56]. In April 2017, a first remote-controlled merchant ship was developed and demonstrated jointly by Rolls-Royce and Svitzer [57].

China pays more and more attentions to the technologies of unmanned ships. In terms of scientific research, China developed the unmanned ship "Tianxiang No. 1" in 2008 which was the first unmanned meteorological detection ship in the world at the time [58]. It can monitor the environment and warn the disasters of oceans and large lakes in time. In December 2015, China classification society issued the code for smart ships, which took effect on March 1, 2016 [59]. In 2017, China classification society, Zhuhai municipal government, and Wuhan university of technology jointly launched a first small unmanned intelligent cargo ship project in China [60]. This small unmanned cargo ship named "JinDou cloud" is 500 ton of displacement with a hull length of 50 meters. It is powered by electricity, has an endurance of up to 500 nautical miles, and can sail autonomously and berth automatically [61].

C. CORE TECHNOLOGIES OF UNMANNED SHIPS

Various technologies are required to make unmanned ships a reality. Fortunately, we already have most of these technologies [56]. Especially, autonomous ship controllers and engine systems are not a hypothesis but a reality [97]. We summarize and introduce currently available core technologies of unmanned ships in detail below, shown in Table 4. Specially, state-of-the-art of core technologies and applications for unmanned ships is presented in detail in Table 5, including intelligent awareness, communication, and sensor data fusion.

Navigation is the process of planning and executing the operations from one position to another by accurately determining the position and the speed of an object relative to a known reference point [68]. There are four main methods widely used in autonomous navigation, including a satellite navigation, a dead reckoning navigation, an inertial navigation, and a multiple sensor navigation.

• Satellite navigation uses navigation satellites to navigate and locate ships on the sea. GPS navigation, Galileo navigation, and Beidou navigation are the common satellite navigation [69]. Satellite navigation has the advantages

TABLE 4. Currently available core technologies for unmanned ships [10], [11], [56], [62].

Core technologies	Description
gyroscope	Gyroscope transmits some "ship sense", which indicates average and peak roll and heave, vibrations dangerous to the hull's structural integrity driving from slamming and taking green water on deck, to a remote officer of the watch
intelligent awareness	Intelligent awareness rely on advanced sensors on board which sense for hazard detection/avoidance and situational awareness, e.g., automated surveillance cameras, accelerator sensors, etc. It is explained in detail in Section IV.
sensor fusion	This is used to monitor, evaluate, and process individual sensor data to produce an improved output of sensors. It is explained in detail in Section IV.
route planning	Route planning of manned ship is laid by the second officer. Route planning of unmanned ship is explained in detail in this section.
avoid collision	Avoid collision of unmanned ship is explained in detail in this section.
communication	Communication of unmanned ships include HF (high frequency), VHF, satellite, 3G/4G/5G, etc. It is explained in detail in Section IV.
navigation	Unmanned ships mainly rely on GPS and DGPS for navigation.
autonomous navigation	It is explained in detail in this section.
energy efficiency control	This relies on advanced energy control algorithms.
status monitoring	Ship navigation data monitoring.
fault diagnosis	In the case of possible fault of a ship, fault diagnosis system can be used to diagnose fault timely and effectively to prevent and reduce the occurrence of accidents.
cargo supervision system	Cargo supervision system monitor standard for Identification, positioning, and management of cargo.
emergency response system	Emergency respond system can make real-time transmission of the situation of a ship when it is in danger or threatened by safety. Shore control center use computer software assesses stability, residual strength, and oil overflow of the ship.
lidar (electro-optical system)	Light detection and ranging calculate the relative distance between the target [63], according to the time of laser re-entry after encountering the obstacle.
Radar	It is explained in detail in Section II.
dynamic positioning	Dynamic positioning (DP) refers to the technology of maintaining the position of a ship or floating platform by means of an automatic propeller controlled by a shipboard computer without anchoring.
GDPS	It is explained in detail in Section II.
strain gauge	Strain gauge monitor strain of ships and equipment during navigation.
ECDIS	Electronic chart display and information system provides digital chart and its application system to assist decision-making for navigation. It is explained in detail in Section II.
iceberg tracking	Iceberg can also be detected by radar and sonar onboard.
transponder	Automatic Identification System installed on the ship can achieve automatic response.
remote human vision	SCC is important guaranteeing for the safe navigation of unmanned ships. SCC monitors unmanned ships in real time and intervenes when danger or human intervention is required.
artificial intelligence	Deep learning has a good application in path tracking and path planning of unmanned ships
cloud computing	Unmanned ships send status data and voyage data to the cloud through the network.Cloud servers use these data for training and learning to provide auxiliary support for decision-making.
edge computing	In the process of navigation, the ship needs to calculate real-time information such as speed, course and the coordinates of obstacles, so as to perform the next step of operation, deceleration, berthing and turning. This complex set of calculations must be performed in real time and with low latency. If the data is processed on a cloud server, any delay in data transfer can lead to disaster.

of all-weather, high precision, and no accumulation of errors and the disadvantages as lacking altitude information and easy to be disturbed [70].

- Dead reckoning is a method to obtain the track and the position of a ship without the aid of external navigational objects. It relies on compass, log, ship's control factors, and air currents [71].
- An inertial navigation system uses the inertial element (accelerometer) to measure the acceleration of the carrier itself [72]. It integrates and operates the acceleration to get the speed and position of the carrier. A practical application is presented in the paper [73].

Inertial navigation's advantages include complete autonomy, complete motion parameters, and strong antiinterference. Its disadvantages include error accumulation and high cost [70].

• A multi sensor navigation system uses radar information and integrates the data which is collected by other sensors. An application is presented in the paper [74].

Ships can use Synthetic Aperture Radar (SAR) image data to track icebergs. SAR combines high resolution with day-and-night weather-independent capability, detects and tracks most icebergs which are about 20 meters (65 feet) in size [75].

TABLE 5.	State-of-the-art of	f core technologies	and applications f	or unmanned sh	ips [64]–[67].
----------	---------------------	---------------------	--------------------	----------------	----------------

Core technolo- gies	Explanation	State-of-the-art
Intelligent awareness	An intelligent awareness system of an unmanned ship provides real time emer- gency warning, object detection, target recognition of severe environment and weather, assists the crew to operate safely, and reduce the risk of human error.	An intelligent awareness system enhances the sit- uational awareness of ship surrounding. It needs advanced camera and sensor technologies support. At present, intelligent awareness system of a ship mainly consists of radar, camera, Lidar, antenna, PTZ (Pan/Tilt/Zoom), etc.
Communication	In the process of remote monitoring, the communication system only needs to send the ship's position, course, speed, and ship status flags. All flags are represented by 2 bytes. In the process of status investigation, SCC needs to query the detailed status information of the ship and the surrounding situation. The communication system needs to transmit radar images, IR and/or video images, HDTV images, and automation. The image is about 300 to 2600 kByte. In the process of indirectly remote control, SCC may need to adjust position or speed waypoints of the voyage. The communication requirements are as for status investigation. In the process of directly remote control, the SCC operators may need to access rudder and thruster controls. Although this is not very demanding in terms of bandwidth, the sensors on the ship require higher bandwidth.	The main communication channels when a ship is close to the coast include VHF, GMDSS, 3G/4G/5G, and etc. The communication has disadvantages of sig- nal transit in short range. The communication chan- nels of deep sea mainly use satellite communication with disadvantages of high cost, and not all satellite communication systems support global coverage and signal transit disturbance.
Sensor data fu- sion	There are three basic processing architectures including direct fusion of sensor data, fusion of feature vectors, and processing of each sensor to achieve high- level inferences or decision.	There are some widely-spreading data fusion ar- chitectures including Joint Directors of Laborato- ries (JDL), the Luo and Kay architecture, and the Dasarathy's architecture. There are also a series of challenges of data fusion including data imperfec- tion, data inconsistency, data confliction, data align- ment/registration and correlation, data type hetero- geneity, fusion location, and dynamic fusion.

A transponder is a wireless communications device usually attached to a satellite [76]. The transponder is mainly used as the Automatic Identification System (AIS) transmitter/receiver in the ocean. An AIS can effectively reduce ship collision accidents. The AIS can be connected with radar, Automatic Radar Plotting Aid (ARPA), Electronic Chart Display and Information System (ECDIS), Vessel Trafc Service (VTS), and other terminal equipments [77]. According to International convention of International Maritime Organization (IMO) for the safety of life at sea, international voyaging ships with a gross tonnage of over 300 tons and all passenger ships must be equipped with an AIS [78].

The report [10] introduces that route planning is one of the key technologies of unmanned ships. The purpose of route planning algorithms is to plan a collision-free feasible path from the starting point to the target point. Next, We summarize and explain algorithms for route planning in detail below, shown in Table 6. In the paper [79], route planning for unmanned ships is classified to global route planning based on marine environmental information and local route planning based on sensor information.

- Global route planning obtains static obstacle information of the sea area through the electronic chart and uses an A star algorithm, a distance optimization Dijkstra algorithm, a genetic algorithm, and an artificial potential field method to search for collision free path from beginning to end. The description of the above algorithms are shown in Table 6.
- Local route planning is to determine the optimal route from the current route point to next by real-time

detection of the surrounding environment and obstacles through sensors when the ship's environmental information is complete or partly unknown [80].

Collision avoidance at sea is fundamental to ensure the safe navigation of unmanned ships. A collision avoidance system includes obstacle detection, tracking, motion estimation, advanced algorithms, and other modules. At present, the technologies of obstacle detection, tracking, and motion estimation are very mature. We focus on collision avoidance algorithms for unmanned ships in this paper. There are many collision dangers when unmanned ships in voyage encounter other ships or other obstacles. obstacles include dynamic ships and static obstacles which are not shown on shipborne Electronic Chart Display and Information System (ECDIS). Generally, collision avoidance algorithms of unmanned ships initially compute a global optimal path. If a moving ship is detected in near proximity by sensors, the algorithms re-plan the next subpath in real-time to avoid the approaching ship [97]. In addition to above factors, the effect of sea wind, wave, and ocean current on autonomous navigation of unmanned ships should be considered in actual navigation conditions. Most of these algorithms are based on the existing navigation rules and real-time navigation information to simulate human piloting cognitive ability [98]. The route planning algorithms listed in Table 6 can avoid unmanned ships collisions. We also summarize other algorithms for collision avoidance, as shown in Table 7. Unmanned ship navigation may use efficient real-time intelligent collision avoidance algorithms to ensure navigation safety.

Classification	Algorithm / Method	Description	Reference
global route planning based on marine environmental information	A star algorithm	The A star algorithm performs a Best-first search of the most probable paths leading to the goal.	[81]
	Distance optimization di- jkstra algorithm	Dijkstra algorithm is an algorithm for calculating single source shortest path in a weighted directed graph [82].	[83]
	Genetic algorithm	The genetic algorithm simulates the mechanisms of natural selection and of evolution theory to search the optimal solution [84].	[85]
	Artificial potential field method	Artificial potential field method is a typical online path algo- rithm. It uses the idea of "water flows to low places" and can naturally understand the generation law of vehicle routing.	[86]
	Ant Colony (ACO)	ACO algorithm is a bionic optimization algorithm to simulate the intelligent behavior of ants.It has the advantages of robust- ness and distributed computing mechanism.	[87], [88]
local route planning based on sensor information	Flower Pollination Algo- rithm (FPA)	FPA is a new metaheuristic algorithm. It was developed from the characteristic of the biological flower pollination in flower- ing plant [89].	[90]
	Rapidly-exploring Random Trees algorithm (RRT)	RRT is a random data structure.It rapidly expand like a tree to explore most of the area of space and find a viable path.	[91]
	Fuzzy logic	Fuzzy logic has robustness and can avoid the characteristics of traditional algorithms, which are sensitive to positioning accuracy and highly dependent on environmental information	[92], [93]
	Neural network	Neural network has the abilities of large-scale parallel process- ing of data and strong fusion of knowledge and can implement efficient and intelligent path planning.	[94]
	Other		[95], [96]

TABLE 6. A summary of route planning algorithms of unmanned ships.

There are many algorithms which can find the optimal routes and avoid collisions for unmanned ships. We analyze and compare typical algorithms of route planning (Table 6) and collision avoidance (Table 7) based on [16]–[19].

- An A* is an optimization algorithm with a modified best-first-search (BFS) strategy which uses heuristic cost estimations [99]. At the beginning of path planning, an A* algorithm [100] chooses a pixel with the lowest F(n) cost adjacent to the start pixel in a sonar image. The vital key to determine which pixels are chosen by the equation F(n) = G(n) + H(n) [101]. G(n) is the actual cost from the begin pixel to a given pixel(n). H(n) is the estimated cost from the given pixel(n) to the goal pixel. Optimal paths are generated by repeatedly going to goal and selected the pixel with the lowest F(n) cost [101].
- A Genetic Algorithm (GA) is a stochastic optimization algorithm based on principles of natural selection and genetics [102]. The key of applying genetic algorithm to route planning of unmanned ships is a genetic representation of the path. The optimal path of unmanned ships is obtained by calculating fitness functions [85].
- In the Artificial Potential Field (APF) model, the target which produces the attraction to unmanned ships, obstacles which produce repulsion to unmanned ships [103]. The next waypoint can be computed by the steering angle of unmanned ships, step size, and current position.

An artificial potential algorithm [86] finds out the direction of the resultant force by adding vectors of the forces. The disadvantages of this algorithm include the problem of local optimal solution in applications.

- An Ant Colony Algorithm (ACA) is a probabilistic algorithm that simulates the behaviour of real ants in search of food [104]. An ACA [105] divides maritime space into grids, establishes its node matrix in the navigation area, and calculates the objective function based on the routes found by each ant. The ACA deletes the nodes that are not passed and repeats this process until finding an optimal route of unmanned ships.
- A neural network is inspired by neural network in human brain and used to simulate human thinking. Back Propagation (BP) network is a widely used neural network. The BP network gives a risk degree of collision based on risk factors of ship encounter. After taking collision avoidance actions, the BP network obtains a new risk degree to evaluate the effectiveness of collision avoidance actions. A neural network algorithm [94] has strong learning ability, but it has a disadvantage of poor generalization ability.

The above algorithms and methods can implement route planning of unmanned ships and avoid collision in navigation. One or several algorithms are often used to enable unmanned ships to navigate in the optimal path and avoid collision with obstacles in practical applications.

TABLE 7. A summary of collision avoidance methods of unmanned ships.

Algorithm / Method	Description	Reference
Support Vector Machine(SVM)	SVM is a model which can find an optimal hyperplane to meet the requirements of classification. The hyperplane equation is defined as $\omega \cdot x + b = 0$, where ω is the normal of the hyperplane and b is the distance from the hyperplane to the origin point [106]. The characteristic parameter of multi-sensor is the environment state vector as the input of SVM, and the ship's next heading angle movement as the output of SVM. A SVM system implements real-time control for the local path planning of a ship [107]. SVM can be combined with other machine learning methods to monitor learning models, analyze collision avoidance data, and identify patterns to improve the accuracy of collision avoidance decisions.	[108], [106], [107].
Finite State Machine (FSM)	FSM is a timing Machine and divides a complex problem into several simple parts to deal with. FDM is a model which describes the sequence of states that an object experiences in its life cycle and how it responds to various events. It can be reduced to 4 elements including current state, condition, action, and secondary state [109]: the current state refers to the navigation state in real-time; the condition is used to determine the state of migration in the next cycle, such as angle deviation between real-time heading and destination; the action is related navigation operations; the second state is the next state compared with the current state.	[110], [109]
Line-of-Sight (LOS)	The core idea of LOS guidance is that the ship converges to a constant LOS heading angle between the ship and the target [97]. The LOS angle (ψ_{los}) is calculated in terms of the current (x,y) and LOS coordinates (x_{los}, y_{los}) . The equation is $\psi_{los} = (y_{los} - y)/(x_{los} - x)$.	[111], [112], [97]
Multi-sensor data fusion (MSDF)	MSDF refers to the collection, processing and collaborative combination of data collected by various knowledge sources and sensors to provide auxiliary decision-making [113]. Various estimation algorithms available for MSDF include Kalman filtering based approach, hybrid multi-sensor data fusion, fuzzy logic based adaptive Kalman filter, and crisp decision algorithm [114].	[115], [113], [114]

IV. TECHNOLOGIES AND APPLICATIONS FOR FUTURE UNMANNED MERCHANT SHIPS

In the field of ships, GPS, AIS, ARPA (Automatic Radar Plotting Aid), ECDIS, and other technologies are used to enable unmanned ships to sail with information and intelligence. However, current technologies cannot fully guarantee the automatic perception and autonomous safe navigation of unmanned merchant ships. Unmanned ships monitor navigation environment, ship status, and working status of equipment through a variety of sensors and intelligent sensing equipments. The various intelligent devices on unmanned ships are integrated into a network through the ship buses or wireless networks to communicate with each other and be gathered to make decisions. Finally, the data are sent to Shore Control Center (SCC) through satellites, the Internet, and 3G/4G/5G mobile communication for monitoring status of unmanned ships in real time. Unmanned ships may need human intervention and remote management in some complex environments and dangerous situations. Next, we introduce IoT technologies, intelligent awareness, data fusion, communication, and current status of IoT for unmanned ships in detail.

A. INTERNET OF THINGS

Internet of Things (IOTs) are described as follows: every thing in the physical world is defined as a unique identity, which connects to the Internet for communicating, collecting, and exchanging data [116]. It is a bridge between the physical world and the virtual world. IoT is a world wide

VOLUME 8, 2020

network of interconnected objects based on standard communication protocols [117]. It extends to the exchange of information and communication between the things in the physical world [118]. Although IOT applications in unmanned ships are still in the early stage as we survey in later subsections, we believe that IOT will be a very promising technology to apply to the unmanned ships in the near future. Therefore, next, we follow architecture, communication standards, security, essential technologies of IoT for introduction.

1) IOT ARCHITECTURE

The typical IoT architecture is 3-layer architecture [119], [120], consisting of a perception layer, a network layer, and an application layer. The authors in [121] propose a 4-layer architecture because the functions and operations in the network and application layers are complex. Based on this concept, the service layer (also known as the interface layer or the middleware layer) is developed between the network layer and the application layer. The service layer has various functions, including service discovery, service composition, service management, and service interfaces. The authors in [122], [123] divide the structure of IoT into 5 layers, including the perception layer, the network layer, the middleware layer, the application layer, the business layer. The IoT architecture is shown as Fig. 6.

2) WIRELESS COMMUNICATION STANDARDS

IoT communication technologies connect different things to provide intelligent services. When the heterogeneous things

TABLE 8. Wireless communication technologies [119], [124]–[126].

Technology	Protocols	Frequency Band	Data Rate	Range
Wifi	802.11	2.4 GHz and 5 GHz	1 Mb/s to 6.75 G/s	20 to 100 M
LoRa		868 and 900 MHz	0.3 Kb/s to 50 Kb/s	20 M
Z-Wave		908.42 MHz to 868.42 MHz	9.6 Kb/s	30 to 100 M
6LoWPAN	802.15.4	868/915 MHz		
NFC		13.56 MHz	424 Kb/s	
WiMax	802.16	2.4 Ghz-5 GHz	1 Mb/s-6.75 Gb/s	
BlueTooth	802.15.1	2.4 GHz	1 Mb/s to 24 Mb/s	8 to 10 M
Zigbee	802.15.4	868 MHz, 915 MHz and 2.4 GHz	250 Kb/s	1Km
Sigfox		868 MHz	4-12 B/h	30-50 KM
LR-WPAN	802.15.4	868/915 GHZ and 2.4 GHz	40 Kb/s to 250 Kb/s	15 KM
DASH7		433 MHz	167 Kb/s	2 KM



FIGURE 6. IoT architecture [119]-[123].

are connected to the same or different network, the communication needs to comply with the relevant protocols to exchange data. We summarize the major wireless communication technologies of IoT, shown in Table 8.

3) IOT SECURITY

The IoT is applied to the most important areas of the global economy, including smart gird, smart city, transportation, and health care. Therefore, security and privacy are the most important issues in IoT. There are six security features of IoT, including confidentiality, integrity, availability, identification and authentication, privacy, and trust [121]. Based on the paper [121], we summarize the security challenges in the perception layer, the network layer, and the application layer, shown in Table 9.

4) ESSENTIAL IOT TECHNOLOGIES

The Radio frequency identification (RFID) tags, sensors/ actuators, Wireless Sensor Network (WSN) [127], and communication technologies are the foundation of IoT [128]. Several enabling technologies of IoT are introduced as follows:

• RFID technology was invented in 1948, but the largescale use of RFID in commercial products appeared in

TABLE 9. IoT Security [121].

Layer	Security	
	Node Capture Attacks	
	Malicious code Injection Attacks	
	False Data Injection Attacks	
Perception Layer	Replay Attacks (or Freshness Attacks)	
	Cryptanalysis and Side Channel Attacks	
	Eavesdropping and Interference	
	Sleep Deprivation Attacks	
	Denial-of-Service (DoS) Attacks	
	Spoofing Attacks	
	Sinkhole Attacks	
Network I aver	Wormhole Attacks	
Network Layer	Man in the Middle Attack	
	Routing Information Attacks	
	Sybil Attacks	
	Unauthorized Access	
	Phishing Attack	
Application Layer	Malicious Virus/Worm	
	Malicious Scripts	

the 1980s [129], [130]. It is a technology for automated identification of objects and people. An RFID system includes transponders or RFID tags, a transceiver or RFID reader, and a back-end database [131], [132]. RFID technology has two distinct advantages as unique identification and automation. In addition to acting as a pointer to a database entry, unique identifiers in RFID tags contribute to containing a large amount of transaction histories for individual items [133], [134]. RFID tags are readable at rates of hundreds per second without line-of-sight contact and without precise positioning.

• WSN is a multi-hop self-organizing network system which consists of a large number of micro sensor nodes deployed in the monitoring area and forms through wireless communication [135]–[137]. Compared with



FIGURE 7. Multi-sensors information fusion processing model [146].

traditional technologies used for monitoring, WSN entails low installation and maintenance costs and enables the development of distributed collaborative applications [138]–[140].

- LPWAN (Low-Power Wide-Area Network) technologies are very popular for IOTs. The most popular LPWAN technologies are LoRa, Sigfox, and NB-IoT. These technologies are well suited for IoT applications that require little information to be transmitted over long distances. LoraWan is a derivative of chirp spread spectrum (CSS) modulation, which can operate the remote communication link in the unlicensed frequency band below 1GHz frequency band below 1GHz [141]. Sigfox uses Ultra Narrow Band (UNB) patented technology and deploys its proprietary base stations in several countries in the unauthorized sub GHz ISM frequency band (such as 868mhz in Europe, 915MHz in North America, and 433MHz in Asia) [142]. Sigfox has the advantages of low power consumption, high receiver sensitivity, and low cost. NB-IoT uses single carrier frequency division multiple access (SC-FDMA). A NB-IoT carrier uses twelve 15 kHz subcarriers for a total of 180 kHz [143]. It is widely used in LTE design.
- Fog computing is a distributed collaborative architecture [144]. In this pattern, data, processing, and applications are concentrated on devices at the edge of the network. Fog computing is characterized by low delay and position perception. It can be widely used in smart vehicle, smart grids, and smart cities, etc.

The IoT applications can be deployed various industries, such as smart city, smart grid, smart transportation, smart home, environmental monitoring, healthcare service, workplace and home support, security, and surveillance [12]. We focus on IoT applications of unmanned ships discussed in Section IV-F.

B. INTELLIGENT AWARENESS

Unmanned ships can collect data of surrounding environments, internal states of ship, and working states of equipment by applying IOT and tracking technology, such as RFID systems and intelligent sensors. We take temperature sensors [145] as an example since the exhaust temperature of engines is an extremely important parameter for the operation of unmanned ships. The exhaust temperature of engines can be obtained in real time through the temperature sensors to ensure the reliability and safety of the unmanned ships' operation. The temperature sensors carry out data acquisition for the ambient temperature signal after digital-to-analog conversion processing and data filtering and transformation. The data are sent to the central console for processing and storage through the radio frequency module. Generally, the data collected include ship sailing status and equipment data information, such as ship speed, heading, position, compass information, indoor temperature of main engines, liquid level, fuel consumption, remaining oil, concentration of warehouse smoke, temperature, and humidity, etc. Ships obtain the channel, the position of the surrounding ships, and navigation information using radar, compass, AIS, ECDIS, GPS, HD camera, and other electronic equipments. Decision-making systems of unmanned ships through a comprehensive analysis of the information make decisions for ship navigation. In addition to collecting internal status data of ships, unmanned ships also need to obtain external environment data, such as wind, wave, current, and other meteorological information, as well as water depth, velocity, and other environmental information, through sensors to ensure safe navigation.

C. MULTI-SOURCE HETEROGENEOUS DATA FUSION

Because of different types of sensors and/or data from different sensor/RFID systems, there may be redundant, conflicting, and missing. Data must be handled with uncertainties of the sources and inconsistency of multi-sensor data before using them at decision processes. This increases the credibility of the data and the ability to make decisions based on the data. Moreover, environments that ships sail at sea can be very harsh. Various sensors on board are susceptible to aging devices, bad weather, limited vision of targets, and other factors, resulting in decreased accuracy and unstable measurement data. Therefore, data fusion must be carried out to give a sufficient information support to the decision maker, so that the decision maker can make a correct decision and judgment on the state of the ship. Fig. 7 shows

The following is a specific application to illustrate. The

a multi-sensor information fusion processing model proposed by the US data fusion working group [146]. The model is composed of information source, source pretreatment, target assessment, situation assessment, threat assessment, overall process assessment, man-machine interface, and database management system module. At present, the computational intelligence methods mainly used in multi-sensor fusion include fuzzy set theory, neural network, rough set theory, wavelet analysis theory, and support vector machine, etc. By collecting multi-sensor data with fusion technologies, the unmanned ship system realizes automatic route planning, collision avoidance, and safe navigation on the water surface. The authors in [79], [147]–[151] study and improve data fusion algorithms and architecture, and improve the stability of the fusion algorithms.

D. COMMUNICATION

We classify communication of unmanned ships into internal and external communications. The internal communications of unmanned ships include CAN (Controller Area Network) bus, WSNs (Wireless Sensor Networks), Zigbee, etc. The external communications of unmanned ships include 3G/4G/5G, satellite communication, VHF, etc. The authors in [152] summarizes various use cases for communications of unmanned ship navigation, automation, and safety. Unmanned ships can adopt different communication technologies according to different requirements and types of communications.

In terms of internal communications of unmanned ships, conventional onboard monitoring systems transmit and collect data through wired networks, LANs (local area networks), fiber optic networks, and CAN buses. Unmanned ships can collect and integrate the information provided by the existing electronic navigation equipments and sensors through CAN buses or LANs to provide decision support for platform software and applications of shipboard monitoring system based on CAN bus technology.

TABLE 10	. Wireless	communication	onboard	[155]–[157].
----------	------------	---------------	---------	--------------

Position	Environment	Transmission
Cabin	watertight doors	No
Cabin	wood doors	Yes
Stairs	no sensor nodes deployed in the stairway	No
	sensor nodes deployed in the stairway	Yes

There are many difficulties in transmitting data using traditional wired methods, such as the need to drill holes in the ship's metal plates for network wiring, resulting in additional costs and waste. WSNs have a number of advantages such as low cost, large range, flexible layout, and mobile support. WSNs can conveniently realize intelligent sensing and monitoring of ship environments. We summarize wireless communication onboard, shown in Table 10 [155]–[157]. authors in [158] measure the received powers of radio wave propagation on board. The measurement results show that the wireless communication depends on the distance between the transmitting antenna and the receiving antenna and position of the deck stairs. Wireless communication onboard between two Sensor Nodes (SNs) is impossible if adjacent rooms are separated by metal bulkheads and watertight doors. Each metal room is a separate network zone. Therefore, we need two types of wireless communication, including intra-zone communication and inter-zone communication. Border Nodes (BNs) are installed in every door in each room each room. Communication between SN and BN is a intra-zone communication. Gateway Nodes (GNs) are the bridges between the host system and the sensor network, which is connected to an Ethernet of the ship. The authors in [159]-[161] deploy WSNs on ships and simulated environments, and conduct relevant tests to verify that WSNs have good reliability and connectivity. In wireless communication technology for collecting sensor data, 2.4Ghz zigbee is usually adopted for wireless communication. Zigbee has the advantages of low power consumption, low cost, short delay, large network capacity, high security, and reliable data transmission. A Zigbee system is a small network with a maximum of 255 nodes per sink node. It is composed of multiple sensors, whose data are sent to the router, which then sends the acquired data to the sink node, which can access to satellite communication and the Internet.



FIGURE 8. The stair of the ship "Yukun".

Next, we take the ship "Yukun" as an example to introduce the wireless communication between the adjacent decks. The ship "Yukun" has a seven-floor deck, including a tween deck, a main deck, a promenade deck, a boat deck, a captain deck, a navigation deck, and a compass deck. The stairs from the tween deck to the main deck have a height of about 3.5 m and have steel walls with a thickness of over 20 mm, shown as Fig. 8. Therefore, wireless communication transmission between the tween deck and the main deck is very difficult. We deploy routers near the stairs, which are located in the stairway to ensure the connectivity between decks, shown



FIGURE 9. The locations of the sensor nodes, sink node, and router.

as Fig. 9, similar to [160]. The green line represents the link between the tween deck and the main deck. Meanwhile, routers are deployed near doors, which can solve the wireless communication difficulty to pass through metal walls and watertight doors.

On the other hand, in terms of external communications of unmanned ships, traditional ship communication system only relies on Automatic Identification System (AIS) to provide low data services such as position, course, heading, destination, tonnage, speed, etc [163]. When unmanned ships are closer to shore, 3G, 4G, and 5G mobile networks can be used for communication. The shore control centers (SCCs) can view the real-time data and historical data of unmanned ship operation, and make decisions based on the data. On the other hand, if unmanned ships are far away from shore, the choice of communication is limited. Satellite communication is commonly used by ships in high sea. Current satellite communication remains very expensive and impractical for most small to mid-sized ships [163]. Due to factors such as obstacles and weather, the transmission of satellite communications data may be delayed and real-time services cannot be provided [164]. Satellite communication may be not the best solution for communication of unmanned ships. In general, lower frequency bands radio is a good choice. The authors in [152], [165], [166] list the bands of the radio spectrum as defined by the International Telecommunications Union (ITU) [165]. IoT on unmanned ships is depended on Very High Frequency (VHF), Ultra High Frequency (UHF), and High Frequency (HF) radio. VHF and HF bands can provide ranges of anything between 40 nautical miles to worldwide coverage [167]. The authors in [162] propose a new communication architecture which combine Ship Ad-hoc Networks (SANETs) and multiple sensors and communicate over marine VHF radio, shown in Fig. 10.



FIGURE 10. IoMat cartography network [162].

E. IoT FOR E-NAVIGATION

E-navigation implements collection, integration, exchange, display, and analysis of marine information on board and in shore control center through electronic means, so as to enhance the navigation and related services of berths for safety of the sea and the protection of the marine environment [168]. E-navigation integrates existing and new navigation tools in a holistic and systematic way. The role of E-navigation is explained as follows:

- E-navigation can improve decision support to improve navigation safety. The crew on board and managers in Shore Control Center (SCC) can acquire the relevant and real-time information of ships. E-navigation provides automatic indication and alarm to reduce human error. Moreover, E-navigation promotes the use of Electronic Chart Display and Information System (ECDIS) of consistent quality.
- E-navigation improves navigation safety, reduces the risk of collision, grounding, and the resulting leakage and pollution. At the same time, E-navigation can adopt the best route and speed to reduce emissions.
- E-navigation provides automated and standardized procedures to reduce management costs. E-navigation integration of existing systems effectively uses new equipments and meets user requirements.

IoT for E-navigation is shown Fig. 11 [169]–[172]. The perception system of e-navigation uses IOT technologies, such as sensors, RFID, WSN, zigbee, to obtain real-time information. After data fusion and processing, the information of GPS, electronic charts, and other onboard equipment is combined with an expert system and artificial intelligence technology for decision assistance and early warning.

F. IoT FOR UNMANNED SHIPS

IoT plays an important role in manufacturing, navigation, and maintenance of unmanned ships. Applications of IoT on ships simplify and optimize aspects of unmanned ship



FIGURE 11. A novel IoT for E-navigation architecture [169]–[172].

operations at sea, in ports, and in formation, including ships tracking, maintenance, safety, etc. The report [173] presents key opportunities of IoT, such as transport and logistics, administrative costs, advances safety, fuel consumption, and efficiency. Therefore, we summarize applications of IoT for unmanned ships as follows:

(1) Advanced sensors and IT technology can implement real-time management and monitoring of unmanned ship navigation, as well as autonomous navigation by unmanned ship management systems. Ship owners can monitor the ship's status in real time and make decisions based on current and historical data, enabling them to operate more efficiently and save time and fuel. Furthermore, in the era of IoT, smart sensors can also be used for ships tracking or predictive maintenance on ships. By embedding important parts and equipment of the ship into sensors, various dynamic information can be monitored in real time and transmitted to Shore Control Centers (SCCs) through the network to detect the health status of the ship in real time. The data can be used as reference information for ship repair, real-time detection of defective parts, timely issuing maintenance warning, and building a ship IOT system. The authors in [174] present that IoT can cover the stages of design or production of the ship. At the same time, sensors are installed in the components that need monitoring to obtain the information about the life of the ships. In the paper [175], the authors propose an application of IoT on the ship to monitor the health in realtime sensing. The article [176] points out that IoT can enable ship owners and managers to find potential failures through real-time monitoring of ship equipment. Several applications of IoT for ship safety are shown in Table 11.

(2) According to statistics, fuel cost is usually the most expensive item in ship operation cost, accounting about 50% of ship operation voyage cost [177], [178]. Sensors and

amounts of fuel and lower carbon emission. On September 4, 2018 [179], International Maritime Satellite Organization (INMARSAT) and the company Danelec Marine jointly released fleet data, which enable owners and managers to quickly and easily identify equipment problems and failures, monitor ship performance, and fuel efficiency. Furthermore, IoT technologies were used to sense obstacles around the ship, environment, location, ocean current, and weather information. IoT provides data support for path planning to make ships run more efficiently, saving time and fuel. Moreover, the data collected by sensors are also the basis of collision avoidance and navigation decisions to avoid ship collision and leakage of environmental pollution. With today's ships becoming highly sensory, shipping management companies and owners have more data than ever about day-to-day operations of ships. Several applications of IoT for fuel consumption are shown in Table 11. (3) IoT technology can reduce operational costs, increase efficiency, and minimize losses due to human errors. For example, IoT allows optimal placement of containers on ships

monitoring equipment onboard collect the ships' perfor-

mance data and send them to Shore Control Center (SCC),

which provides guidance in planning the most energy-

efficient routes. After determining the optimal route, speed, and engine configuration, ships can save considerable

example, IoT allows optimal placement of containers on ships to reduce offloading times. IoT can track exactly when ships arrive at ports, reducing the cost of waiting for trucks to move containers to the next station. Moreover, the IoT technology can save massive amount of human cost. With the RFID technology, active electronic tags are installed on the ships entering and leaving channel. The tags can store individual information related to the ships or cargo information. A remote reader is installed on the channel, which connects to a data server through the standard serial port RS232/485.

TABLE 11. Several applications of IoT for shipping [180], [181].	

Aspects	Country	Company	Application	Detail
Safety	American	ioCurrents	An IoT processing unit capable of measuring all of the different parameters in a ship's engine.	It can enhance navigation safety and provide potential faults in time.
	United King- dom	Green Sea Guard	A gas analysis tools measure ship's engine exhaust gas.	The gas monitor gives engineers an early warning when a ship's engine is not operating correctly. This gives engineers an early warning.
	American	Augury	Halo sensors	The sensor measure ship's engine vibration, sound, and temperature. The diagnostics platform analyzes raw data through machine learning to provide real-time alerts and health reports about in-depth view of engine health.
Fuel consumption	Finland	Wärtsilä	Dynamic Maintenance Planning (DMP)	DMP analyzes the operating conditions of the ship engines and sends it via satellite to maximize intervals between maintenance periods, thus reducing administrative costs and carbon emission.
Cost efficiency	American	Carnival Corpora- tion	An IoT-based personalized digital concierge service.	The service makes use of near field communication (NFC) and Blue- tooth Low Energy (BLE) technology to implement streamlined embark- ing, locking cabin doors, anytime-anywhere gaming, and easy payment.
	United King- dom	ZS Wellness	Wearable health tracking system	The data collected can be used to create personalised wellness plans for individual crew members.
Economy and trade	American	Parsyl	A sensor which can be fitted to an individual pallet or package.	The sensor can measure temperature, humidity, light, impact, and GPS position. Moreover, the data from the sensor can across land, sea, and air. It can be used for cargo status monitoring in ocean transportation.
	Denmark	Maersk	Remote Container Management (RCM) system	The system monitors and sends data such as temperature and location to the cloud. RCM can reduce spoilage of goods, such as fresh produce.

In the area where it is difficult to place wiring, a wireless data transmission module (GPRS/CDMA) can be linked to the data server for real-time communication. The reader can identify and read the electronic tags installed on the ships at a distance of up to 80 meters, so as to implement non-stop inspection. In addition, ship owners or managers need to monitor the health of the crew in real time. The crew members wear wearable devices while on board to collect health data. The data help managers in SCC to better plan welfare provision for those on board. Several applications of IoT for cost efficiency are shown in Table 11.

(4) For cargo ships, managers need to keep track of ships' position, status, and cargo environment information at all times. Sensors on ships collect data and send them to SCC in real time via communication. Meanwhile, sensors can track temperature, humidity, light, impact, and GPS position, etc. On the other hand, for passenger ships, passengers on board can wear smart wearable devices to implement navigation, door access, payment, and other functions. Several applications of IoT for commercial trade are shown in Table 11.

(5) We think that there are also three type applications of IoT in unmanned ships including front-end data acquisition, data communication, and data storage and analysis processing. The front-end data acquisition is to collect the data of navigation parameters, navigation attitude, and surrounding environment from various types of sensors on ships. The collected data is transmitted by IoT communication technologies, including ZigBee, RFID, Bluetooth, WSN, or LPWAN. Finally, the collected data are processed and stored through cloud computing, fog computing, and cloud storage to provide decision support for the safe navigation of ships.

V. FUTURE RESEARCH DIRECTIONS

Unmanned merchant ships have the advantages of autonomous navigation, fast speed, strong mobility, good concealment, and low cost, but there are potential safety hazards such as collision, grounding, and security. For example, unmanned merchant ships cannot be converted to manual steering in case of automatic rudder failure. If there is any equipment component that needs to be replaced or repaired during the voyage of an unmanned ship, the manual treatment cannot be carried out in time, and this will endanger the voyage of the unmanned ship. In case of fire, leakage or other dangerous situations, no personnel can be organized for temporary and effective disposal. The problem of the endurance of unmanned ship is that when an unmanned ship needs energy supply, it cannot immediately learned about the unmanned ship in relation to port state legislation, and cannot automatically contact the nearby port to carry out the port supply and continue the voyage mission after supply. These problems are the urgent problems to be solved for future unmanned ships. The authors in [182] show that unmanned merchant ships have serious consequences due to non-navigational accidents (such as fire, structural damage caused by ship loss) compared with traditional merchant ships. Next, we introduce some future research directions of unmanned merchant ships from several aspects: law, communication, IoT Applications, security, onboard intelligent equipment, automatic control, Artificial Intelligence and Big Data, Ship Industry 4.0, and smart ocean.

A. IMPROVEMENT OF LAWS AND REGULATIONS CONCERNING UNMANNED SHIPS

If an unmanned ship without a captain and a crew onboard, is it still a ship? If yes, whether such an unmanned ship is governed by maritime law? If no, are there new laws that applies to unmanned ships? At present, relevant laws on unmanned ships are not perfect at present. It is one of the biggest challenges for the development of unmanned ships in the future. We summarize rules of law, including crew, navigation, rights and duties, shown as follows:

- As unmanned ships no longer need a captain or a crew, it seems that rules of law about these people will inevitably lose all relevance. These laws [183] include Manila amendment, MLC (Maritime Labour Convention), SOLAS (International Convention for Safety of Life at Sea), STCW (International Convention on Standards of Training, Certification and Watchkeeping for Seafarers), and captain's law, etc. On the other hand, there are many managers in Shore Control Center (SCC). Under special circumstances, unmanned aerial vehicles need to be controlled remotely by shore-based managers. Whether do these managers need to be restricted by relevant laws?
- International Regulations for Preventing Collisions at Sea (COLREGs) rules specify which ship is responsible for giving way to other ships and which side of the "stand-on" ship can maneuver in the case of crossing, overtaking, and head-on traffic [184]. Several researchers study the concept of COLREGS-based navigation of unmanned ships [185], [186].
- Several relevant laws [183], [187] stipulate rights and duties in connection with international shipping, including UNCLOS (United Nations Convention on the Law of the Sea), LOSC (Law of the Sea Convention), and maritime law, etc. Several researchers discuss whether the laws are applicable to unmanned merchant ships [188], [189].

The law of unmanned ship should take into consideration the navigation characteristics and functions of unmanned ships, and formulate laws and regulations fully applicable to unmanned ships, so as to facilitate the further development of unmanned ships.

B. COMMUNICATION

When unmanned ships are sailing in the high sea, 3G/4G network cannot be covered and satellite communication is expensive. Although the ships do not need the continuous control of the shore-based control center, the network security and data transmission stability of communication between the ship and the shore-based control center are very important during autonomous navigation. We have summarize following research, which can promote the development of unmanned ships. Firstly, when the ship is sailing in the deep sea, the network cannot be covered. The ship can only rely on satellite communication. However, satellite communication is expensive, which cannot guarantee the speed and stability of network transmission. For example, when the ship "Yukun" is sailing in the deep sea, it reports current position through satellite communication once a day. We believe that

decrease of satellite communication cost will be beneficial to the development of unmanned ships in the future. Secondly, new communication technologies are needed with wider coverage and more convenient communications for unmanned ships at deep sea.

C. CYBER SECURITY

Communication/IoT security has become a concern for shipping. We have to consider specific risks when IoT is deployed for the unmanned ships. The use of technical and functional indicators enables the Shore Control Centers (SCCs) to monitor the status of unmanned ships with minimal operator and communications system load. Moreover, under the background of remote operation of unmanned ships, concerns on cyber security are further increased. Skilful and capable hackers may gain access to the ship control system through the network and implement the purpose of controlling the ship and changing its operation according to the destination set by the hackers. Several researchers study the concept of cyber security of unmanned ships [190], [191]. In our opinion, cyber security is one of the important issues affecting the future development of unmanned ships. In the future, a dedicated framework for specific systems, such as railway, smart grids, and nuclear power plants, will be developed to secure unmanned ships navigation. In addition to the cyber security of communication, the future unmanned merchant ships also have physical security issues, such as:

- If some hostile people (such as pirates) or ships approach or board unmanned ships, they deliberately destroy the communication equipment or interfere with the communication signal onboard. This will cause unmanned ships to lose control and be in danger. It is a problem that must be solved before the actual operations of unmanned merchant ships in the future.
- If the communication equipment is completely enclosed, it can play a protective role, but whether the communication signal can be transmitted normally needs further research and testing.

D. IOT APPLICATIONS ON SHIPS

Although a lot of IoT applications on ships have been mentioned in section IV-F, we think that there will be more iot applications on unmanned ships in the future. At present, equipments on the ship are not modular enough, and the procedures for updating the equipments are quite complex. Future equipments of unmanned ships may be modular and updated with plug-and-play functions. This requires the support of automation, control, communication, and computer technologies. Meanwhile, the development of various types of sensors can promote the development of unmanned ships, such as sensor life, data stability, and power supply, etc. Future IoT applications on ships digitize supply chains, track logistics operations, accurately predict ship arrival times, optimize navigation routes, forecast demand, and simulate complex supply networks. These future applications will greatly improve operation efficiency, improve operation control, increase maintenance reliability, and optimize operation cost.

E. INTELLIGENT AUXILIARY EQUIPMENT

Due to the high electronicalization of unmanned ships, the damage of general electronic devices will cause fatal threat to the safe navigation of unmanned ships. Therefore, unmanned ships should be equipped with related intelligent robots for daily maintenance and repair of unmanned ships in case of problems. Experts in shore-based control center make diagnosis based on the data from unmanned ships. The intelligent robots on the unmanned ship maintain and repair unmanned ships according to the solution from the shorebased control center. Experts monitor the scene of maintenance is transmitted to the shore-based control center in real time.

F. AUTOMATIC CONTROL

The shipboard equipment and navigation instruments introduced in Section II-C can use the IOT technology to connect these non automatic, semi-automatic, and fully automatic equipment to realize automatic monitoring and control in the future. Autopilot is the most typical automatic control equipment onboard. At present, many researchers study this subject. In addition, some carry out a series of researches on the next generation intelligent rudder for unmanned ships navigation [192], [193].

G. ARTIFICIAL INTELLIGENCE AND BIG DATA

Automation systems on most ships collect a large mount of data. Generally, there are hundreds of such data input or output points on a common ship. The number of modern ships may be tens of thousands. These data have three main problems. First, most of the data come from relatively independent and closed systems. Second, the quality of data cannot be guaranteed. Third, the amount of data is huge. Integrated Bridge System (IBS) mentioned in Section II-C integrates navigation system and automation system into one platform, which is convenient for comprehensive analysis and provides decision support for ship navigation. In the future, big data technology will not only store massive data, but also analyze, process, retrieve, and mine these data. Through AI (Artificial Intelligence) related technologies, such as DLS (Deep Learning System) and BLS (Broad Learning System), people and software provide guarantee for the navigation of unmanned merchant ships.

H. SHIP INDUSTRY 4.0

Ship Industry 4.0 (SI4) is a collaborative network integrating IOT, service network, data, and human. It integrates the real space and virtual space based on the virtual production system (CPPS) [194]. SI4 includes smart ship design, smart ship manufacture, and smart ship operation processes. It implements of ships industry from the design of improving energy efficiency to more intelligent and smart operation. Ship manufacture is a highly complex process involving many disciplines. Ship design is an important part before ship manufacture. Ship design generally includes four stages: conceptual design, contract design, preliminary design, and detailed design [195]. In the future, SI4 will be fully automated and directly connected to manufacturing and operating systems, designed according to market trends and customer feedback.

I. SMART OCEAN

So far, we know a relatively small area of the world's oceans. Ships are affected by environment, meteorological conditions, deep-sea communication infrastructure, and other factors. Smart ocean system collects ocean data through shore control center, island station, port, lighthouse, buoy, submarine buoy, Unmanned Aerial Vehicle (UAV), remote sensing satellite, communication satellite, etc. These data are transmitted to big data cloud server through high-speed network, providing solutions for marine ecological protection, marine prediction and disaster reduction, ship navigation, and marine engineering. In the future, unmanned commercial ships need more intelligent equipments to provide information technology support for safe navigation, resource discovery, and environmental protection.

VI. CONCLUSION

Unmanned merchant ships have the ability of autonomous planning, autonomous navigation, and autonomous environment perception. Shore-based control center can carry out real-time intervention control for unmanned ships. In the future, unmanned merchant ships will play a more important role in ocean freight transportation and military affairs. Future research of unmanned merchant ships needs more powerful sensors, perfect unmanned ship related laws and regulations, and a wider and safe network coverage.

This paper introduces the mechanisms of a merchant ship and the duties of the personnel on board from the aspects of structure, principle of ship handing, and navigation instruments. In the future, electric ships may become a better solution for unmanned ships.

We categorize Unmanned Surface Vehicles (USVs) into small, medium, large, and extra large by displacement. We focus on large and extra large displacement USVs as unmanned ships in this paper. We summarizes classification of unmanned ships as automated manned ships, remotecontrol ships with crew on board, remote-control ships without crew on board, and completely unmanned automatic ships according to the level of automation. We list in detail the economic, social and safety benefits of unmanned ships. We summarize the development and the state of art of unmanned ships by studying the existing unmanned ships papers and introduction. We summarize the core technologies of unmanned ships, such as intelligent awareness, sensor fusion, collision avoidance, path planning, and communication, etc. We focus on the classification and introduction of collision avoidance and path planning algorithms, compare the advantages and disadvantages of the algorithms.

We summarizes the support of IOT technologies for unmanned ships, such as intelligent awareness, sensor fusion, and communication, etc. Specially, we summarize IoT For e-navigation architecture. Furthermore, We look forward to the future of IoT for unmanned ships.

We believe that many technological breakthroughs are needed if unmanned ships implement autonomous navigation without intervention or control in the future. But unmanned ships will greatly save human and financial resources and must be the future development trend.

We realize that both unmanned ships and IOT for shipping are still in their early stages although they will have great potentials in the future. Shipping companies are reluctant to reveal their detail designs of their early products except some news reports while academia has fewer solid research publications. We hope that our paper stimulates more research in these areas so that we can see more published research papers in these areas in the near future.

REFERENCES

- Baidu Baike. Merchant Ship. Accessed: Oct. 10, 2018. [Online]. Available: https://baike.baidu.com/item/%E5%95%86%E8%88%B9/ 10753371
- [2] P. Kaluza, A. Kölzsch, M. T. Gastner, and B. Blasius, "The complex network of global cargo ship movements," *J. Roy. Soc. Interface*, vol. 7, no. 48, pp. 1093–1103, Jul. 2010.
- [3] Wikipedia. Merchant Ship. Accessed: Oct. 10, 2018. [Online]. Available: https://en.wikipedia.org/wiki/Merchant_ship
- [4] T. Porathe, "Remote monitoring and control of unmanned vessels— The MUNIN shore control centre," in *Proc. 13th Int. Conf. Comput. Appl. Inf. Technol. Maritime Industries (COMPIT)*, 2014, pp. 460–467. Accessed: Nov. 10, 2019. [Online]. Available: http://publications.lib. chalmers.se/records/fulltext/198197/local_198197.pdf
- [5] Wikipedia. Unmanned Surface Vehicle. Accessed: Apr. 4, 2019. [Online]. Available: https://en.wikipedia.org/wiki/Unmanned_surface_vehicle
- [6] O. J. Rødseth and H. C. Burmeister, "Developments toward the unmanned ship," in *Proc. Int. Symp. Inf. Ships ISIS*, vol. 201, 2012, pp. 30–31.
- [7] International Union of Marine Insurance. *Iumi Stats Report 2018*. Accessed: Oct. 20, 2018. [Online]. Available: https://iumi.com/ document/view/IUMI_Stats_Report_2018_5c62966a7392d.pdf
- [8] HypoVereinsbank. Trendstudie Green Shipping. Accessed: Nov. 20, 2018. https://www.forschungsinformationssystem.de/servlet/is/332401/HVB% 20Studie_Green_Shipping_dt.pdf?command=downloadContent& filename=HVB%20Studie_Green_Shipping_dt.pdf
- [9] Newstrong Technology Company. (Jan. 2019). Unmanned Ship Has Obvious Advantages Over Traditional Ships. Accessed: Oct. 18, 2019.
 [Online]. Available: http://www.nstrong.cn/display/221735.html
- [10] Eworldship. Seven Key Technologies for Smart Ship. Accessd: Dec. 22, 2019. [Online]. Available: http://www. eworldship.com/html/2019/ship_inside_and_outside_0510/149239.html
- [11] Asv Global. (Sep. 2017). Three Vital Elements That Will Make Autonomous Shipping a Reality. [Online]. Available: https://www. roboticstomorrow.com/article/2017/09/the-three-vital-elements-thatwill-make-autonomous-shipping-a-reality-/10696
- [12] L. Da Xu, W. He, and S. Li, "Internet of Things in industries: A survey," *IEEE Trans. Ind. Informat.*, vol. 10, no. 4, pp. 2233–2243, Nov. 2014.
- [13] K. Rose, S. Eldridge, and L. Chapin, "The Internet of Things: An overview," *The Internet Soc. (ISOC)*, vol. 80, p. 1–50, 2015.
- [14] A. A. Fröhlich, "Smartdata: An Iot-ready Api for sensor networks," Int. J. Sens. Netw., vol. 28, no. 3, pp. 202–210, 2018.
- [15] G. Tuna, "Clustering-based energy-efficient routing approach for underwater wireless sensor networks," *Int. J. Sensor Netw.*, vol. 27, no. 1, pp. 26–36, 2018.

- [16] G. L. Zhang, M. X. Hu, J. F. Chai, L. Zhao, and T. Yu, "Summary of path planning algorithm and its application," *Modern Machinery*, no. 5, pp. 85–90, 2011.
- [17] CSDN. (Oct. 2016). Advantages and Disadvantages of SVM. Accessed: Oct. 8, 2019. [Online]. Available: https://blog.csdn.net/zm1_1zm/ article/details/52752214
- [18] J. Kim, S. Kim, and Y. Choo, "Stealth path planning for a high speed torpedo-shaped autonomous underwater vehicle to approach a target ship," *Cyber-Phys. Syst.*, vol. 4, no. 1, pp. 1–16, Jan. 2018.
- [19] H. Mousazadeh, H. Jafarbiglu, H. Abdolmaleki, E. Omrani, F. Monhaseri, M.-R. Abdollahzadeh, A. Mohammadi-Aghdam, A. Kiapei, Y. Salmani-Zakaria, and A. Makhsoos, "Developing a navigation, guidance and obstacle avoidance algorithm for an unmanned surface vehicle (USV) by algorithms fusion," *Ocean Eng.*, vol. 159, pp. 56–65, Jul. 2018.
- [20] Y. X. Jin and S. C. Wu, *Ship Structure and Equipment*. Beijing, China: China Communications Press, 2012.
- [21] Z. J. Guan and T. Liu, Navigation instrument (Book 1: Navigation Equipment for Ships). Dalian, China: Dalian Maritime Univ., Press, 2009.
- [22] W. Li, *Ship Structure and Equipment*. Dalian, China: Dalian Maritime Univ., Press, Aug. 2008.
- [23] Boe Marine. (Jan. 2018). Do you Need Radar. Accessed: Dec. 13, 2018. [Online]. Available: https://www.boemarine.com/blog/post/do-you-need-radar/
- [24] International Hydrographic Organization. (2018). Electronic Navigational Charts (ENCS) & Electronic Chart Display and Information System (ECDIS). Accessed: Mar. 4, 2019. [Online]. Available: http://www. iho.int/srv1/index.php?option=com_content&view=category&id=72& lang=en&Itemid=287
- [25] D. J. Wright and D. J. Barlett, Marine and Coastal Geographical Information System. Boca Raton, FL, USA: CRC Press, 2000.
- [26] B. Wenku. (Jul. 2018). The Duty of Crew. Accessed: Aug. 16, 2019. [Online]. Available: https://wenku.baidu.com/view/ bbd8a158842458fb770bf78a6529647d27283473.html
- [27] V. Bertram, Unmanned surface vehicles—A survey, vol. 1, Copenhagen, Denmark: Skibsteknisk Selskab, 2008, pp. 1–14.
- [28] H. N. H. Oh, A. Tsourdos, and A. Savvaris, "Development of collision avoidance algorithms for the C-Enduro USV," *IFAC Proc. Volumes*, vol. 47, no. 3, pp. 12174–12181, 2014.
- [29] M. Caccia, M. Bibuli, R. Bono, and G. Bruzzone, "Basic navigation, guidance and control of an unmanned surface vehicle," *Auto. Robots*, vol. 25, no. 4, pp. 349–365, Nov. 2008.
- [30] M. Bibuli, G. Bruzzone, M. Caccia, and L. Lapierre, "Path-following algorithms and experiments for an unmanned surface vehicle," *J. Field Robot.*, vol. 26, no. 8, pp. 669–688, Aug. 2009.
- [31] Ø. J. Rødseth and Å. Tjora, "A system architecture for an unmanned ship," in Proc. 13th Int. Conf. Comput. IT Appl. Maritime Industries (COMPIT), 2014, pp. 1–13.
- [32] Ø. J. Rødseth and H. Nordahl. (Mar. 2018). Definitions for Autonomous Merchant Ships. Norwegian Forum for Unmanned Ships, Version 1.0. Accessed: Oct. 10, 2017. [Online]. Available: http://nfas.autonomousship.org/resources.html
- [33] Xin De Maritime. (May 2018). Imo Began to Study the Regulations of Unmanned Ships. Accessed: Apr. 14, 2019. [Online]. Available: http://www.ship.sh/news_detail.php?nid=29420
- [34] MŪNIN. (2016). About MUNN -Maritime Unmanned Navigation through Intelligence in Networks. Accessed: Apr. 8, 2019. http://www. unmanned-ship.org/munin/about/
- [35] Waterborne. About Waterborne. Accessed: Nov. 20, 2018. [Online]. Available: https://www.waterborne.eu/about/about-waterborne/
- [36] D. L. Qiao, J. Hou, and F. Xue, "Design and realization of unmanned surface ship intelligent navigation control system based on Internet of Things," *Ship Sci. Technol.*, vol. 23, pp. 149–152, 2017.
- [37] Microdis. Sara-n2—The First lte cat.nb1 (Narrow band iot) Modules. Accessed: Jun. 4, 2019. [Online]. Available: https://www.microdis.net/ news/0/4720/sara-n2-the-first-lte-catnb1-narrow-band-iot-modules.html
- [38] U. Blox. Sara-n2 Series. Accessed: Aug. 3, 2019. [Online]. Available: https://www.u-blox.com/en/product/sara-n2-series
- [39] Bloomberg. (Jul. 2014)Rolls-Royce Drone Ships Challenge \$ 375 Billion Industry. Accessed: Sep. 9, 2019. [Online]. Available: https://www. questia.com/article/1P2-35740733/rolls-royce-drone-ships-challenge-375-billion-industry
- [40] L. Kretschmann, H.-C. Burmeister, and C. Jahn, "Analyzing the economic benefit of unmanned autonomous ships: An exploratory costcomparison between an autonomous and a conventional bulk carrier," *Res. Transp. Bus. Manage.*, vol. 25, pp. 76–86, Dec. 2017.

IEEEAccess

- [41] Sina Blog. Psychological Analysis of Young Sailors Engage in Navigation. Accessed: Apr. 23, 2019. [Online]. Available: http://blog.sina. com.cn/s/blog_482f832a0102wk61.html
- [42] L. Y. Tang. (Feb. 2014). Rolls-Royce Unmanned Ship: It's Time for the Pirates to Find Another Job. Accessed: Oct. 8, 2019. [Online]. Available: https://www.leiphone.com/news/201406/rolls-royce-ship.html
- [43] L. Kobyliński, "Smart ships–autonomous or remote controlled," Zeszyty Naukowe Akademii Morskiej w Szczecinie, vol. 53, no. 125, pp. 28–34, 2018, doi: 10.17402/262.
- [44] ModerStyle. (Aug. 2018). The World's First Unmanned Container Ship Has Been Built. Will Pirates and Hackers be in the Way. Accessed: Oct. 8, 2018. [Online]. Available: http://dy.163.com/v2/article/ detail/DPOSC5BE0520KPHJ.html
- [45] S. Corfield and J. Young, "Unmanned surface vehicles-game changing technology for naval operations," *IEE Control Eng. Ser.*, vol. 69, p. 311, Jan. 2006.
- [46] NavalDrones. (2012). An/slq-60 Seafox. Accessed: Oct. 8, 2019. [Online]. Available: http://www.navaldrones.com/seafox.html
- [47] National Defense Magazine. (2005). No Crews Required: Unmanned Vessels Hit the Waves. Accessed: Nov. 3, 2018. [Online]. Available: http://www.nationaldefensemagazine.org/issues/2005/oct/sbno crews.htm
- [48] J. Cao and X. S. Wang, "Development status and future prospects of unmanned ships at home and abroad," *China Ship Survey*, vol. 216, no. 5, pp. 149–152, 2017.
- [49] Inquisitr. (2016). U.S. Pentagon: Come Meet the Sea Hunter Unmanned Vessel. Accessed: Oct. 21, 2018. [Online]. Available: https://www. inquisitr.com/2977895/u-s-pentagon-come-meet-the-sea-hunterunmanned-vessel/amp/
- [50] B. Enderle, T. Yanagihara, M. Suemori, H. Imai, and A. Sato, "Recent developments in a total unmanned integration system," in *Proc. AUVSI* Unmanned Syst. Conf., vol. 9, 2004, p. 9.
- [51] Yamaha. (Oct. 2016). Industrial Drone With Maximum Payload 35Kg,Maximum Altitude 2,800m, Cruising Range of 90km Yamaha Motor: Fazer r G2 Automated Navigation Unmanned Helicopter Showcased at the October, 'Japan International Aerospace Exhibition 2016. Accessed: Jan. 26, 2020. [Online]. Available: https://global.yamahamotor.com/news/2016/1011/fazer_r_g2.html
- [52] A. Kumar and J. Kurmi, "A review on unmanned water surface vehicle," *Int. J. Adv. Res. Comput. Sci.*, vol. 9, no. Special no. 2, p. 95, 2018.
- [53] Defense Update. (Nov. 2006). Stingray—Unmanned Surface Vehicle (USV). Accessed: Mar. 11, 2019. https://defense-update.com/ 20061121_stingray.html
- [54] navaldrones. (2012). Silver Marlin Usv. Accessed: Nov. 9, 2019. [Online]. Available: http://www.navaldrones.com/Silver-Marlin.html
- [55] Dynamical Systems and Ocean Robotics Lab. General Description of a Dsor Lab. Accessed: Oct. 8, 2019. [Online]. Available: http://dsor. isr.ist.utl.pt/introduction/
- [56] International Shipping News. (Oct. 2017). Unmanned Ship: A Dream or Reality. Accessed: Oct. 8, 2019[Online]. Available: https://www. hellenicshippingnews.com/unmanned-ship-a-dream-or-reality/
- [57] World Maritime News. (Mar. 2018). In Depth: Unmanned Ships—Are We There Yet. Accessed: Oct. 8, 2019. [Online]. Available: https:// worldmaritimenews.com/archives/247204/interview-unmanned-shipsare-we-there-vet/
- [58] H. Huang, *Guided Radar Technology*. Beijing, China: Publishing House of Electronics industry, Oct. 2006.
- [59] China Classification Society. (2015). Smart Ship Specification. Tech. Rep. Accessed: Apr. 4, 2019. [Online]. Available: http://www.ccs.org. cn/ccswz/font/fontAction!downloadArticleFile.do? attachId=ff808081511f069e01515b101f0b029a
- [60] Yun Zhou Tech. Jindou Cloud, Yunzhou Intelligent Unmanned Ship. Accessed: Aug. 8, 2019. [Online]. Available: http://www.yunzhoutech.com/Products/detail/id/50.html
- [61] Sohu. (Dec. 2017). China's First 500-Ton Unmanned Cargo Ship Project has Been Launched. Accessed: Aug. 8, 2019. [Online]. Available: http://www.sohu.com/a/209347220_276266
- [62] MEMS. (Oct. 2017). The Driverless Trend is Here, and the Lidar Industry is Gaining Momentum. Accessed: Oct. 8, 2019. http://www.sohu. com/a/196584634 256868
- [63] J. Zheng, S. Yang, X. Wang, X. Xia, Y. Xiao, and T. Li, "A decision tree based road recognition approach using roadside fixed 3D LiDAR sensors," *IEEE Access*, vol. 7, pp. 53878–53890, 2019.
- [64] O. J. Rodseth, B. Kvamstad, T. Porathe, and H.-C. Burmeister, "Communication architecture for an unmanned merchant ship," in *Proc. MTS/IEEE OCEANS Bergen*, Jun. 2013, pp. 1–9.

- [65] R.-R. Plc. (2018). Intelligent Awareness Enhancing Safety Through Intelligent Data Fusion. Accessed: Mar. 2, 2020. [Online]. Available: https:// www.ferryshippingsummit.com/wp-content/uploads/2018/03/Rolls-Royce.pdf
- [66] M. Liggins II, D. Hall, and J. Llinas, Handbook of Multisensor Data Fusion: Theory and Practice. Boca Raton, FL, USA: CRC Press, 2017.
- [67] T. Meng, X. Jing, Z. Yan, and W. Pedrycz, "A survey on machine learning for data fusion," *Inf. Fusion*, vol. 57, pp. 115–129, May 2020.
- [68] J. Farrell and M. Barth, *The Global Positioning System and Inertial Navigation*. New York, NY, USA: McGraw-Hill, 1999, vol. 61.
- [69] Baidu Baike. Satellite Navigation. Accessed: Apr. 4, 2019. [Online]. Available: https://baike.baidu.com/item/%E5%8D%AB%E6%98%9F% E5%AF%BC%E8%88%AA/722964?fr=aladdin
- [70] Baidu Baike. Modular Navigator. Accessed: Apr. 4, 2019. [Online]. Available: https://baike.baidu.com/item/%E7%BB%84%E5%90%88% E5%BC%8F%E5%AF%BC%E8%88%AA%E4%BB%AA
- [71] Encyclopaedia Britannica. Dead Reckoning Navigation. Accessed: Apr. 4, 2019. [Online]. Available: https://www.britannica. com/technology/dead-reckoning-navigation
- [72] Baidu Baike. inertial Navigation. Accessed: Oct. 8, 2019. [Online]. Available: https://baike.baidu.com/item/%E6%83%AF%E6% 80%A7%E5%AF%BC%E8%88%AA/5951188?fr=aladdin
- [73] B. E. Bona and R. J. Smay, "Optimum reset of Ship's inertial navigation system," *IEEE Trans. Aerosp. Electron. Syst.*, vol. AES-2, no. 4, pp. 409–414, Jul. 1966.
- [74] A. Stateczny and W. Kazimierski, "Sensor data fusion in inland navigation," in *Proc. 14th Int. Radar Symp. (IRS)*, vol. 1, Jun. 2013, pp. 264–269.
- [75] Encyclopaedia Britannica. Iceberg Detection, Tracking, and Management. Accessed: Oct. 8, 2019. [Online]. Available: https://www. britannica.com/science/iceberg/Iceberg-detection-tracking-andmanagement
- [76] Webopedia. Transponder. Accessed: Oct. 8, 2019. [Online]. Available: https://www.webopedia.com/TERM/T/transponder.html
- [77] Baidu Baike. Automatic Identification System of Ships. Accessed: Oct. 8, 2019. [Online]. Available: https://baike.baidu.com/item/%E8% 88%B9%E8%88%B6%E8%87%AA%E5%8A%A8%E8%AF%86% E5%88%AB%E7%B3%BB%E7%BB%9F/1873702?fr=aladdin
- [78] Wikipedia. Transponder. Accessed: Dec. 18, 2018. [Online]. Available: https://en.wikipedia.org/wiki/Transponder
- [79] Y. Chen, Y. X. Hu, Y. N. Liu, and X. D. Zhu, "Processing and fusion for multi-sensor data," J. Jilin Univ.(Sci. Ed.), vol. 56, no. 5, pp. 136–144, 2018.
- [80] Baidu Baike. Route Planning. Accessed: Oct. 8, 2019. [Online]. Available: https://baike.baidu.com/item/%E8%B7%AF%E5%BE%84% E8%A7%84%E5%88%92/8638339?fr=aladdin
- [81] F. Duchoň, A. Babinec, M. Kajan, P. Beňo, M. Florek, T. Fico, and L. Jurišica, "Path planning with modified a star algorithm for a mobile robot," *Procedia Eng.*, vol. 96, pp. 59–69, Jan. 2014.
- [82] W. M. W. Y. F. Y. Miao and Q. Hongbing, "A new path planning strategy of a data collection problemutilising multi-mobile nodes in wireless sensor networks," *Int. J. Sens. Netw.*, vol. 29, no. 3, pp. 192–202, 2019.
- [83] J. Y. Zhuang, L. Wan, Y. L. Liao, and H. B. Sun, "Global path planning of unmanned surface vehicle based on electronic chart," *Comput. Sci.*, vol. 38, no. 9, pp. 211–214, 2011.
- [84] M.-C. Tsou, S.-L. Kao, and C.-M. Su, "Decision support from genetic algorithms for ship collision avoidance route planning and alerts," *J. Navigat.*, vol. 63, no. 1, pp. 167–182, Jan. 2010.
- [85] H. Kim, S.-H. Kim, M. Jeon, J. Kim, S. Song, and K.-J. Paik, "A study on path optimization method of an unmanned surface vehicle under environmental loads using genetic algorithm," *Ocean Eng.*, vol. 142, pp. 616–624, Sep. 2017.
- [86] K. Liu, Y. H. Zhang, and J. Ren, "Path planning algorithm for unmanned ship based on improved artificial potential field method," *Natural Sci. J. Hainan Univ.*, vol. 34, no. 2, pp. 99–104, 2016.
- [87] C. H. Song, "Global path planning method for USV system based on improved ant colony algorithm," in *Applied Mechanics and Materials*, vol. 568. Zurich Switzerland: Trans Tech Publ, 2014, pp. 785–788.
- [88] W. Yuan-hui and C. Cen, "Research on optimal planning method of USV for complex obstacles," in *Proc. IEEE Int. Conf. Mechatronics Autom.*, Aug. 2016, pp. 2507–2511.

- [89] S. Griffith. (2018). Flower Pollination Algorithm. Accessed: Aug. 26, 2019. [Online]. Available: https://slideplayer.com/slide/ 15036940/
- [90] Y. Zhou and R. Wang, "An improved flower pollination algorithm for optimal unmanned undersea vehicle path planning problem," *Int. J. Pattern Recognit. Artif. Intell.*, vol. 30, no. 04, May 2016, Art. no. 1659010.
- [91] J. Y. Zhuang, L. Zhang, H. B. Sun, and Y. M. Su, "Improved rapidlyexploring random tree algorithm application in unmanned surface vehicle local path planning," *J. Harbin Inst. Technol.*, vol. 47, no. 1, pp. 112–117, 2015.
- [92] S.-M. Lee, K.-Y. Kwon, and J. Joh, "A fuzzy logic for autonomous navigation of marine vehicles satisfying COLREG guidelines," *Int. J. Control, Autom., Syst.*, vol. 2, no. 2, pp. 171–181, 2004.
- [93] D. Mu, Y. Zhao, G. Wang, Y. Fan, and Y. Bai, "Course control of USV based on fuzzy adaptive guide control," in *Proc. Chin. Control Decis. Conf. (CCDC)*, Yinchuan, China, May 2016, pp. 6433–6437.
- [94] R. Glasius, A. Komoda, and S. C. A. M. Gielen, "Neural network dynamics for path planning and obstacle avoidance," *Neural Netw.*, vol. 8, no. 1, pp. 125–133, Jan. 1995.
- [95] J. Y. Zhuang, Y. M. Su, Y. L. Liao, and H. B. Sun, "Unmanned surface vehicle local path planning based on marine radar," *J. Shanghai Jiaotong Univ.*, vol. 46, no. 9, pp. 1371–1375, 2012.
- [96] C. X. Hou, J. S. Xu, and R. W. Yang, "Local path planning and Path following control for inland water USV," *Ship Eng.*, to be published.
- [97] S. Campbell, W. Naeem, and G. W. Irwin, "A review on improving the autonomy of unmanned surface vehicles through intelligent collision avoidance manoeuvres," *Annu. Rev. Control*, vol. 36, no. 2, pp. 267–283, Dec. 2012.
- [98] G. Robert, J. Hockey, A. Healey, M. Crawshaw, D. G. Wastell, and J. Sauer, "Cognitive demands of collision avoidance in simulated ship control," *Hum. Factors, J. Hum. Factors Ergonom. Soc.*, vol. 45, no. 2, pp. 252–265, Jun. 2003.
- [99] R. Dechter and J. Pearl, "Generalized best-first search strategies and the optimality of A*," J. ACM, vol. 32, no. 3, pp. 505–536, Jul. 1985.
- [100] T. Phanthong, T. Maki, T. Ura, T. Sakamaki, and P. Aiyarak, "Application of A* algorithm for real-time path re-planning of an unmanned surface vehicle avoiding underwater obstacles," *J. Mar. Sci. Appl.*, vol. 13, no. 1, pp. 105–116, Mar. 2014.
- [101] P. Lester. (2005). A* Pathfinding for Beginners. GameDev Web-Site. Accessed: Feb. 8, 2009. [Online]. Available: http://www.gamedev. net/reference/articles/article2003.asp
- [102] D. E. Goldberg, "Genetic and evolutionary algorithms come of age," *Commun. ACM*, vol. 37, no. 3, pp. 113–119, Mar. 1994.
- [103] S. Xie, P. Wu, Y. Peng, J. Luo, D. Qu, Q. Li, and J. Gu, "The obstacle avoidance planning of USV based on improved artificial potential field," in *Proc. IEEE Int. Conf. Inf. Autom. (ICIA)*, Jul. 2014, pp. 746–751.
- Baidu Baike. Ant Colony Algorithm. Accessed: Oct. 8, 2019.
 [Online]. Available: https://baike.baidu.com/item/%E8%9A% 81%E7%BE%A4%E7%AE%97%E6%B3%95/9646604?fr=aladdin
- [105] M.-C. Tsou and H.-C. Cheng, "An ant colony algorithm for efficient ship routing," *Polish Maritime Res.*, vol. 20, no. 3, pp. 28–38, Sep. 2013.
- [106] K. Zheng, Y. Chen, Y. Jiang, and S. Qiao, "A SVM based ship collision risk assessment algorithm," *Ocean Eng.*, vol. 202, Apr. 2020, Art. no. 107062.
- [107] J. Tian, M. Gao, and E. Lu, "Dynamic collision avoidance path planning for mobile robot based on multi-sensor data fusion by support vector machine," in *Proc. Int. Conf. Mechatronics Autom.*, Aug. 2007, pp. 2779–2783.
- [108] G. S. Wang, S. Xie, C. G. Liu, X. M. Chu, and Z. L. Li, "Obstacle identification method based on laser radar for inland unmanned vessel," *Opt. Techn.*, vol. 44, no. 5, pp. 602–608, 2018.
- [109] D. Wang, J. Zhang, J. Jin, and X. Mao, "A local steering path planning algorithm based on finite state machine model," *Mar. Sci.*, vol. 42, no. 1, pp. 119–127, 2018.
- [110] D. Wang, J. Zhang, J. C. Jin, and X. P. Mao, "A local steering collision avoidance path planning algorithm based on finite state machine model," *Mar. Sci.*, vol. 42, no. 1, pp. 119–127, 2018.
- [111] S. Moe and K. Y. Pettersen, "Set-based line-of-sight (LOS) path following with collision avoidance for underactuated unmanned surface vessel," in *Proc. 24th Medit. Conf. Control Autom. (MED)*, Jun. 2016, pp. 402–409.

- [112] P. A. Wilson, C. J. Harris, and X. Hong, "A line of sight counteraction navigation algorithm for ship encounter collision avoidance," *J. Navigat.*, vol. 56, no. 1, pp. 111–121, Jan. 2003.
- [113] P. Varshney, "Multisensor data fusion," *Electron. Commun. Eng. J.*, vol. 9, no. 6, pp. 245–253, Jan. 1998.
- [114] T. Xu, R. Sutton, and S. Sharma, "A multi-sensor data fusion navigation system for an unmanned surface vehicle," *Proc. Inst. Mech. Eng., M, J. Eng. Maritime Environ.*, vol. 221, no. 4, pp. 167–182, Dec. 2007.
- [115] J. Kang, M. Jin, J. Park, and D. Park, "A study on application of sensor fusion to collision avoidance system for ships," in *Proc. ICCAS*, Oct. 2010, pp. 1741–1744.
- [116] P. Lynggaard, "Controlling interferences in smart building iot networks using machine learning," *Int. J. Sens. Netw.*, vol. 30, no. 1, pp. 46–55, 2019.
- [117] R. V. Kranenburg. (2008). The Internet of Things: A Critique of Ambient Technology and the all-Seeing Network of Rfid. Amsterdam, The Netherlands. [Online]. Available: https://lib.pyu.edu.vn/bitstream/handle/ 123456789/4449/The-Internet-of-Things-269.pdf?sequence=1&isAllowed=n
- [118] N. A. M. Alduais, J. Abdullah, A. Jamil, and H. Heidari, "APRS: Adaptive real-time payload data reduction scheme for IoT/WSN sensor board with multivariate sensors," *Int. J. Sensor Netw.*, vol. 28, no. 4, pp. 211–229, 2018.
- [119] A. Al-Fuqaha, M. Guizani, M. Mohammadi, M. Aledhari, and M. Ayyash, "Internet of Things: A survey on enabling technologies, protocols, and applications," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 2347–2376, 4th Quart., 2015.
- [120] K. Zhao and L. Ge, "A survey on the Internet of Things security," in *Proc.* 9th Int. Conf. Comput. Intell. Secur., Dec. 2013, pp. 663–667.
- [121] J. Lin, W. Yu, N. Zhang, X. Yang, H. Zhang, and W. Zhao, "A survey on Internet of Things: Architecture, enabling technologies, security and privacy, and applications," *IEEE Internet Things J.*, vol. 4, no. 5, pp. 1125–1142, Oct. 2017.
- [122] S. Kraijak and P. Tuwanut, "A survey on Internet of Things architecture, protocols, possible applications, security, privacy, real-world implementation and future trends," in *Proc. IEEE 16th Int. Conf. Commun. Technol.* (*ICCT*), Oct. 2015, pp. 26–31.
- [123] O. Said and M. Masud, "Towards Internet of Things: Survey and future vision," Int. J. Comput. Netw., vol. 5, no. 1, pp. 1–17, 2013.
- [124] J. Tan and S. G. M. Koo, "A survey of technologies in Internet of Things," in *Proc. IEEE Int. Conf. Distrib. Comput. Sensor Syst.*, May 2014, pp. 269–274.
- [125] G. A. Akpakwu, B. J. Silva, G. P. Hancke, and A. M. Abu-Mahfouz, "A survey on 5G networks for the Internet of Things: Communication technologies and challenges," *IEEE Access*, vol. 6, pp. 3619–3647, 2018.
- [126] P. P. Ray, "A survey on Internet of Things architectures," J. King Saud Univ.-Comput. Inf. Sci., vol. 30, no. 3, pp. 291–319, 2018.
- [127] P. Zhang, M. Abdelaal, and O. Theel, "Quality of service control in proactive wireless sensor networks via lifetime planning," *Int. J. Sens. Netw.*, vol. 26, no. 4, pp. 252–268, 2018.
- [128] R. V. Kranenburg, E. Anzelmo, A. Bassi, D. Caprio, S. Dodson, and M. Ratto, "The Internet of Things," in *Proc. Draft Paper Prepared 1st Berlin Symp. Internet Soc.*, Berlin, Germany, Oct. 2011.
- [129] M. A. Khan, M. Sharma, and B. R. Prabhu, "A survey of rfid tags," Int. J. Recent Trends Eng., vol. 1, no. 4, p. 68, 2009.
- [130] Y. Xiao, S. Yu, K. Wu, Q. Ni, C. Janecek, and J. Nordstad, "Radio frequency identification: Technologies, applications, and research issues," *Wireless Commun. Mobile Comput.*, vol. 7, no. 4, pp. 457–472, 2007.
- [131] P. Peris-Lopez, J. C. Hernandez-Castro, J. M. Estevez-Tapiador, and A. Ribagorda, "Rfid systems: A survey on security threats and proposed solutions," in *Proc. Int. Conf. Pers. Wireless Commun.*, vol. 4217, Albacete, Spain, Sep. 2006, pp. 159–170.
- [132] Y. Xiao, X. Shen, B. O. Sun, and L. Cai, "Security and privacy in RFID and applications in telemedicine," *IEEE Commun. Mag.*, vol. 44, no. 4, pp. 64–72, Apr. 2006.
- [133] L. Zeng, D. Grau, and Y. Xiao, "Assessing the feasibility of passive and BAP RFID communications on construction site scenarios," *IEEE Syst. J.*, vol. 10, no. 4, pp. 1505–1515, Dec. 2016.
- [134] Y. Xiao, Y. Zhang, and X. Liang, "Primate-inspired communication methods for mobile and static sensors and rfid tags," ACM Trans. Auto. Adapt. Syst. (TAAS), vol. 6, no. 4, p. 26, 2011.

- [135] Baidu Baike. Wireless Sensor Network. Accessed: Oct. 8, 2019. [Online]. Available: https://baike.baidu.com/item/WSN%E7%BD% 91%E7%BB%9C/1885875?fr=aladdin
- [136] Y. Xiao, H. Chen, K. Wu, B. Sun, Y. Zhang, X. Sun, and C. Liu, "Coverage and detection of a randomized scheduling algorithm in wireless sensor networks," *IEEE Trans. Comput.*, vol. 59, no. 4, pp. 507–521, Apr. 2010.
- [137] R. Liscano, N. B. Otman, S. S. Heydari, and D. Sharma, "Fixed node assisted collection tree protocol for mobile wireless sensor networks," *Int. J. Sens. Netw.*, vol. 31, no. 3, pp. 133–144, 2019.
- [138] F. Losilla, A.-J. Garcia-Sanchez, F. Garcia-Sanchez, J. Garcia-Haro, and Z. J. Haas, "A comprehensive approach to WSN-based ITS applications: A survey," *Sensors*, vol. 11, no. 11, pp. 10220–10265, Oct. 2011.
- [139] L. Wang and Y. Xiao, "A survey of energy-efficient scheduling mechanisms in sensor networks," *Mobile Netw. Appl.*, vol. 11, no. 5, pp. 723–740, Oct. 2006.
- [140] S. T. P. Pal and C. Kumar, "Clustered heterogeneous wireless sensor network infrastructure for reliable and efficient path planning of mobile nodes in remote area," *Int. J. Sens. Netw.*, vol. 31, no. 4, pp. 199–215, 2019.
- [141] R. S. Sinha, Y. Wei, and S.-H. Hwang, "A survey on LPWA technology: LoRa and NB-IoT," *ICT Express*, vol. 3, no. 1, pp. 14–21, Mar. 2017.
- [142] K. Mekki, E. Bajic, F. Chaxel, and F. Meyer, "Overview of cellular LPWAN technologies for IoT deployment: Sigfox, LoRaWAN, and NB-IoT," in *Proc. IEEE Int. Conf. Pervas. Comput. Commun. Workshops* (*PerCom Workshops*), Mar. 2018, pp. 197–202.
- [143] H. Mroue, A. Nasser, S. Hamrioui, B. Parrein, E. Motta-Cruz, and G. Rouyer, "MAC layer-based evaluation of IoT technologies: LoRa, SigFox and NB-IoT," in *Proc. IEEE Middle East North Afr. Commun. Conf. (MENACOMM)*, Apr. 2018, pp. 1–5.
- [144] W. Han and Y. Xiao, "Edge computing enabled non-technical loss fraud detection for big data security analytic in smart grid," J. Ambient Intell. Humanized Comput., vol. 11, no. 4, pp. 1697–1708, Apr. 2020.
- [145] F. A. Padovani, T. H. McMains, and M. R. Rowlette, "Temperature sensor," U.S. Patent 5 372 427, Dec. 13, 1994. [Online]. Available: https://patents.glgoo.top/patent/US5372427A/en
- [146] F. White, Data Fusion Lexicon: Data Fusion Subpanel of the Joint Directors of Laboratories Technical Panel for C3. San Diego, CA, USA: IEEE Transactions, 1991.
- [147] X. G. Zhang and J. C. Zheng, "A study on the information fusion of the shipborne navigation instruments," *Chin. J. Sci. Instrum.*, vol. 26, no. 3, p. 2004, 2005.
- [148] F. Mazzarella, M. Vespe, and C. Santamaria, "SAR ship detection and self-reporting data fusion based on traffic knowledge," *IEEE Geosci. Remote Sens. Lett.*, vol. 12, no. 8, pp. 1685–1689, Aug. 2015.
- [149] M. Guerriero, P. Willett, S. Coraluppi, and C. Carthel, "Radar/ais data fusion and SAR tasking for maritime surveillance," in *Proc. 11th Int. Conf. Inf. Fusion*, Jun. 2008, pp. 1–5.
- [150] Y. Fischer and A. Bauer, "Object-oriented sensor data fusion for wide maritime surveillance," in *Proc. Int. WaterSide Secur. Conf.*, Nov. 2010, pp. 1–6.
- [151] B. Zhou, L. Xie, and M. Z. Wu, "Key technology of vts and ais information fusion in the Internet of Inland ships," *Ship Sci. Technol.*, vol. 8, no. 16, pp. 124–126, 2016.
- [152] K. Nybom, W. Lund, S. Lafond, J. Lilius, J. Björkqvist, K. Suominen, and K. Tuulos, "IoT at sea," in *Proc. IEEE Int. Symp. Broadband Multimedia Syst. Broadcast. (BMSB)*, Jun. 2018, pp. 1–7.
- [153] M. H. Zhang and X. L. Feng, "Research on the application of can bus in the communication system of marine diesel engine," *Ship Sci. Technol.*, no. 2X, pp. 70–72, 2016.
- [154] H. Jiang, Q. Cai, and J. Zhou, "Design and implementation of general marine engine monitoring system with configurable interface," *Mach. Building Automat.*, no. 3, p. 52, 2017.
- [155] B.-G. Paik, S.-R. Cho, B.-J. Park, D. Lee, B.-D. Bae, and J.-H. Yun, "Characteristics of wireless sensor network for full-scale ship application," *J. Mar. Sci. Technol.*, vol. 14, no. 1, pp. 115–126, Mar. 2009.
- [156] B.-G. Paik, S.-R. Cho, B.-J. Park, D. Lee, J.-H. Yun, and B.-D. Bae, "Employment of wireless sensor networks for full-scale ship application," in *Proc. Int. Conf. Embedded Ubiquitous Comput.* Taipei, Taiwan: Springer, Dec. 2007, pp. 113–122.
- [157] B.-G. Paik, S.-R. Cho, B.-J. Park, D. Lee, and B.-D. Bae, "Development of real-time monitoring system using wired and wireless networks in a full-scale ship," *Int. J. Nav. Archit. Ocean Eng.*, vol. 2, no. 3, pp. 132–138, Sep. 2010.

- [158] E. Mauricio-Hannula, "The future of wireless sensor network on board a ship/vessel," School Eng., Oulu Univ. Appl. Sci., Oulu, Finland, Tech. Rep., 2005. [Online]. Available: http://www.oamk. fi/~karil/mit_studies/wireless_future_seminar/papers2013/final_ paper_mauricio-hannula_elizabeth.pdf
- [159] S. Q. Hu, Y. C. Liu, and F. Y. Zhou, "A study on the supervision network design of engine room based on wsn and fieldbus," *Mar. Electr. Electron. Eng.*, vol. 31, no. 10, pp. 46–48, 2011.
- [160] A. Sarkar, S. Majumdar, and P. P. Bhattacharya, "Estimation of signal strength in a wsn for application in war ship," *Advance Electron. Electr. Eng.*, vol. 3, no. 4, pp. 433–438, 2013.
- [161] H. Kdouh, C. Brousseau, G. Zaharia, G. Grunfeleder, and G. E. Zein, "A realistic experiment of a wireless sensor network on board a vessel," in *Proc. 9th Int. Conf. Commun. (COMM)*, Jun. 2012, pp. 189–192.
- [162] R. Al-Zaidi, J. C. Woods, M. Al-Khalidi, and H. Hu, "Building novel VHF-based wireless sensor networks for the Internet of marine things," *IEEE Sensors J.*, vol. 18, no. 5, pp. 2131–2144, Mar. 2018.
- [163] R. Al-Zaidi, J. Woods, M. Al-Khalidi, and H. Hu, "An IOT-enabled system for marine data aquisition and cartography," *Trans. Netw. Commun.*, vol. 5, p. 53, Feb. 2017.
- [164] B. Hui, K. Jeon, K. Chang, S. Kim, J. Park, and Y. Lim, "Design of radio transmission technologies for VHF band ship ad-hoc network," in *Proc. ICTC*, Sep. 2011, pp. 626–629.
- [165] Nomenclature of the Frequency and Wavelengh Bands Used in Telecommunications, document Recommendation itu-r v.431-8 (08/2015), International Telecommunication Union, Aug. 2015.
- [166] J. A. Stine and D. L. Portigal, "Spectrum 101: An introduction to spectrum management," Mitre Corp Mclean Va Washington C3I DIV, VA, USA, Tech. Rep. MTR 04W0000048, Mar. 2004. [Online]. Available: https://www.mitre.org/sites/default/files/pdf/04_0423.pdf
- [167] S. Lag, P. Andersen, B.-J. Vartdal, and K. E. Knutsen, "Ship connectivity," DNV GL Strategic Res. & Innov. Position Paper, vol. 4, pp. 1–48, 2015.
- [168] D. Patraiko, "The development of e-navigation," *TransNav, Int. J. Mar. Navigat. Saf. od Sea Transp.*, vol. 1, no. 3, pp. 257–260, 2007.
- [169] F. Amato, M. Fiorini, S. Gallone, and G. Golino, "E-navigation and future trend in navigation," *TransNav, Int. J. Mar. Navigat. Saf. Od Sea Transp.*, vol. 5, no. 1, pp. 11–14, 2011.
- [170] C. Wang and G. Peng, "Application of Internet of Things in development of E-navigation architecture," in *Proc. Int. Symp. Comput. Inform.* Paris, France: Atlantis Press, 2015, pp. 579–586.
- [171] C. B. Wang, X. Y. Zhang, and J. J. Li, "Development ideas of unmanned ship navigation technology based on e-navigation," *J. Jimei Univ.(Natural Sci.)*, no. 5, pp. 354–359, 2018.
- [172] K. An, M. of Oceans, and Fisheries, "A study on the improvement of maritime traffic management by introducing e-navigation," *J. Korean Soc. Mar. Environ. Saf.*, vol. 21, no. 2, pp. 164–170, Apr. 2015.
- [173] safety4sea. (Jan. 2018). Iot Opportunities in the Smart Shipping Era. Accessed: Nov. 1, 2019. [Online]. Available: https://safety4sea.com/cmiot-opportunities-in-the-smart-shipping-era/
- [174] Sname. (Oct. 2018). Shipyard 4.0 and Internet of Ships, the Iot Applied to Shipbuilding. Accessed: Oct. 31, 2019. [Online]. Available: https://www. sname.org/westerneurope/events/event-description?CalendarEventKey= fb7375b5-ac9e-4c51-8efc-f5cc5db0eef1&CommunityKey=ecaaf73eafcf-473b-b2cd-6f1cbe6df864&Home=%2Fevents%2Fcalendar
- [175] S. Yang, L. Shi, D. Chen, Y. Dong, and Z. Hu, "Development of ship structure health monitoring system based on IoT technology," in *IOP Conference Series, Earth and Environmental Science*, vol. 69, no. 1. Bristol, United Kingdom: IOP Publishing, 2017, Art. no. 012178.
- [176] W. Hannemann. (Jun. 2019). How lot Can be Used in the Maritime Industry. Accessed: Oct. 31, 2049. [Online]. Available: https://www.dualog. com/blog/how-iot-can-be-used-in-the-maritime-industry
- [177] M. M. Golias, G. K. Saharidis, M. Boile, S. Theofanis, and M. G. Ierapetritou, "The berth allocation problem: Optimizing vessel arrival time," *Maritime Econ. Logistics*, vol. 11, no. 4, pp. 358–377, Dec. 2009.
- [178] M. Stopford, "Maritime economics 3E," in *The Regulation of The Maritime Industry*, 3rd ed. Evanston, IL, USA: Routledge, 2008, doi: 10.4324/9780203891742.
- [179] Inmarsat. Inmarsat Unveils Major New IoT Service for the Shipping Industry. Accessed: Nov. 4, 2019. [Online]. Available: https://www.inmarsat.com/news/inmarsat-unveils-major-new-iotservice-for-the-shipping-industry/
- [180] Thetius. Maritime Applications for IoT. Accessed: Nov. 1, 2019. [Online]. Available: https://thetius.com/maritime-applications-for-iot/

- [181] O. B. Services. (Jan. 2017). IoT on the High Seas. Accessed: Nov. 3, 2019. [Online]. Available: https://www.orange-business.com/en/magazine/ioton-the-high-seas
- [182] K. Wróbel, J. Montewka, and P. Kujala, "Towards the assessment of potential impact of unmanned vessels on maritime transportation safety," *Rel. Eng. Syst. Saf.*, vol. 165, pp. 155–169, Sep. 2017.
- [183] T. Karlis, "Maritime law issues related to the operation of unmanned autonomous cargo ships," WMU J. Maritime Affairs, vol. 17, no. 1, pp. 119–128, Mar. 2018.
- [184] Y. Kuwata, M. T. Wolf, D. Zarzhitsky, and T. L. Huntsberger, "Safe maritime autonomous navigation with COLREGS, using velocity obstacles," *IEEE J. Ocean. Eng.*, vol. 39, no. 1, pp. 110–119, Jan. 2014.
 [185] M. R. Benjamin and J. A. Curcio, "COLREGS-based navigation of
- [185] M. R. Benjamin and J. A. Curcio, "COLREGS-based navigation of autonomous marine vehicles," in *Proc. IEEE/OES Auto. Underwater Vehicles*, Jun. 2004, pp. 32–39.
- [186] W. Naeem, G. W. Irwin, and A. Yang, "COLREGs-based collision avoidance strategies for unmanned surface vehicles," *Mechatronics*, vol. 22, no. 6, pp. 669–678, Sep. 2012.
- [187] H.-C. Burmeister, W. C. Bruhn, O. J. Rødseth, and T. Porathe, "Can unmanned ships improve navigational safety," in *Proc. Transp. Res. Arena (TRA) 5th Conf., Transp. Solutions Res. Deployment*, Paris, France, Apr. 2014, pp. 14–17.
- [188] E. Van Hooydonk, "The law of unmanned merchant shipping-an exploration," *The J. Int. Maritime Law*, vol. 20, no. 3, pp. 403–423, 2014.
- [189] L. Carey, All Hands Off Deck? The Legal Barriers to Autonomous Ships. Singapore: National Univ. of Singapore, Social Science Electronic Publishing, Aug. 2017, doi: 10.2139/ssrn.3025882.
- [190] J. H. Wang and X. P. Wu, "Effectivness analysis of ship communication security equipment based on fuzzy synthetic evaluation," *Ship Sci. Technol.*, vol. 31, no. 4, pp. 103–106, 2009.
- [191] S. K. Katsikas, "Cyber security of the autonomous ship," in Proc. 3rd ACM Workshop Cyber-Phys. Syst. Secur. CPSS, 2017, pp. 55–56.
- [192] S. AloN, S. P. S. Ram, K. Nithin, and A. Kumar, "Efficient rudder control mechanism for unmanned ship navigation," *Int. J. Lastest Res. Eng. Technol.*, vol. 2, pp. 53–56, 2016.
- [193] J. M. Larrazabal and M. S. Peñas, "Intelligent rudder control of an unmanned surface vessel," *Expert Syst. Appl.*, vol. 55, pp. 106–117, Aug. 2016.
- [194] J. Ang, C. Goh, A. Saldivar, and Y. Li, "Energy-efficient through-life smart design, manufacturing and operation of ships in an industry 4.0 environment," *Energies*, vol. 10, no. 5, p. 610, Apr. 2017.
- [195] J. H. Ang, C. Goh, and Y. Li, "Smart design for ships in a smart product through-life and industry 4.0 environment," in *Proc. IEEE Congr. Evol. Comput. (CEC)*, Jul. 2016, pp. 5301–5308.



JIA WANG received the B.S. degree in computer science from Jilin University, Jilin, Changchun, China, in 2002, and the M.E. degree in software engineering from Jilin University, in 2009. He is currently pursuing the Ph.D. degree in traffic information technology and control with Dalian Maritime University. His current research interests include the Internet of Things (IoTs) for unmanned ships and network security of agents.



YANG XIAO (Fellow, IEEE) received the B.S. and M.S. degrees in computational mathematics from Jilin University, Changchun, China, and the M.S. and Ph.D. degrees in computer science and engineering from Wright State University, Dayton, OH, USA. He is currently a Full Professor with the Department of Computer Science, University of Alabama, Tuscaloosa, AL, USA. His current research interests include cyber-physical systems, the Internet of Things, security, wireless networks,

smart grid, and telemedicine. He has published over 290 SCI-indexed journal papers (including over 50 IEEE/ACM transactions papers) and over 250 EI indexed refereed conference papers and book chapters related to these research areas. He was a Voting Member of the IEEE 802.11 Working Group from 2001 to 2004, involving the IEEE 802.11 (WIFI) standardization work. He is an IET Fellow (previously IEE) (FIET). He currently serves as the Editor-in-Chief for *Cyber-Physical Systems* (Journal). He had (s) been an Editorial Board or Associate Editor for 20 international journals, including the IEEE TRANSACTIONS ON CYBERNETICS since 2020, IEEE TRANSACTIONS ON SYSTEMS, MAN, AND CYBERNETICS: SYSTEMS (2014–2015), IEEE TRANSACTIONS ON VEHICULAR TECHNOLOGY (2007–2009), and IEEE COMMUNICATIONS SURVEY AND TUTORIALS (2007–2014). He has served (s) as a Guest Editor for over 20 times for different international journals, including the IEEE NETWORK, IEEE WIRELESS COMMUNICATIONS, and *ACM/Springer Mobile Networks and Applications* (MONET).



TIESHAN LI (Senior Member, IEEE) received the B.S. degree in ocean fisheries engineering from the Ocean University of China, Qingdao, China, in 1992, and the Ph.D. degree in vehicle operation engineering from Dalian Maritime University (DMU), Dalian, China, in 2005. He has been a Full Professor with the School of Automation Engineering, University of Electronic Science and Technology of China, Chengdu, China, since 2019. He was a Lecturer with DMU from 2005 to

2006, where he was an Associate Professor from 2006 to 2011, a Full Professor since 2011, and the Ph.D. Supervisor since 2009. From 2007 to 2010, he was a Postdoctoral Scholar with the School of Naval Architecture, Ocean and Civil Engineering, Shanghai Jiao Tong University, Shanghai, China. From 2008 to 2009 and from 2014 to 2015, he visited the City University of Hong Kong, Hong Kong, as a Senior Research Associate. Since 2013, he has been visiting the University of Macau, Macau, China, as a Visiting Scholar many times. His current research interests include intelligent learning and control for nonlinear systems, and multi-agent systems and their applications to marine vehicle control.



C. L. PHILIP CHEN (Fellow, IEEE) received the M.S. degree from the University of Michigan, Ann Arbor, MI, USA, in 1985, and the Ph.D. degree from Purdue University, West Lafayette, IN, USA, in 1988, both in electrical engineering. He is currently a Chair Professor with the Department of Computer and Information Science, Faculty of Science and Technology, University of Macau, Macau, China. Being a Program Evaluator of the Accreditation Board of Engineering and Technol-

ogy Education (ABET), Baltimore, MD, USA, for computer engineering, electrical engineering, and software engineering programs, he successfully architects the University of Macau's Engineering and Computer Science programs, receiving accreditations from Washington/Seoul Accord through the Hong Kong Institution of Engineers (HKIE), Hong Kong, which is considered as his utmost contribution in engineering/computer science education for Macau as the former Dean of the Faculty. His current research interests include systems, cybernetics, and computational intelligence. He is a Fellow of The American Association for the Advancement of Science, the International Association for Pattern Recognition, the Chinese Association of Automation (CAA), and HKIE. He received the 2016 Outstanding Electrical and Computer Engineers Award from his Ph.D. alma mater, Purdue University. He was the Chair of the TC 9.1 Economic and Business Systems of the International Federation of Automatic Control from 2015 to 2017. He is the Editor-in-Chief of the IEEE TRANSACTIONS ON CYBERNETICS and an Associate Editor of the IEEE TRANSACTIONS ON FUZZY SYSTEMS and the IEEE TRANSACTIONS ON CYBERNETICS. From 2012 to 2013, he was the IEEE Systems, Man, and Cybernetics Society President. He is currently the Vice President of CAA.